Domestic Content in China's Exports and its Distribution by Firm Ownership*

Discussion paper

Hong Ma, Tsinghua University

Zhi Wang, U.S. International Trade Commission

Kunfu Zhu, Academy of Mathematics and System Sciences, CAS

International Conference on the Measurement of International Trade and Economic Globalization Aguascalientes, Mexico, 29 Sep – 1 Oct 2014

United Nations Friends of the Chair Meeting on the Measurement of International Trade and Economic Globalization Aguascalientes, Mexico, 2 Oct 2014

Abstract

Processing trade and foreign-invested enterprises (FIEs) account for a large share of total Chinese exports. In producing exports, they also use imported inputs disproportionately, which complicates the measurement of domestic content embedded in exports and the distribution of income generated from exports. In this paper, we extend the method developed by Koopman, Wang, and Wei (2012) to further distinguish between Chinese exports by FIEs and Chineseowned enterprises (COEs), in addition to processing and normal exports. We propose an accounting framework and a detailed estimation procedure that separately account for the production and trade activities of FIEs and COEs. First, we decompose gross exports into domestic and foreign content by firm types. Then, we estimate factor ownership by firm types based on enterprise surveys. Finally, we compute the distribution of domestic content by factor ownership. Empirical estimation is based on China's 2007 benchmark input-output tables, supplemented by detailed trade and production statistics. Firm heterogeneity within each industry is identified by linking the NBS enterprises survey and the Customs' firm-level trade data. The empirical results from 2007 indicate the following: (1) domestic content accounted for around 59% of total exports; (2) FIEs operating in China created nearly 45% of the domestic content in Chinese exports, whereas processing COEs only contributed by less than 5%; (3) in terms of income distribution, about 52.6% of the value of Chinese exports was captured by foreign factor owners.

^{*} The views in the paper are solely the author's own opinion. It is not meant to represent in anyway the views of the U.S. International Trade Commission or any of its individual Commissioners.

I Introduction

Globalization has made production increasingly fragmented across countries, with roughly two-thirds of world trade accounted by trade in intermediate inputs (Johnson and Noguera, 2012). In particular, outsourcing and foreign direct investment (FDI) to developing countries accelerate the "slicing-up" of the global value chain across national borders, thus widening the gap between the value of gross exports and the actual value added that is created in the exporting country. ¹ Hence, the policies that focus on aggregate trade flow may be very misleading.

In this paper, we propose a new estimation strategy for the domestic content in China's exports by firm types. We focus on China because of its high export/GDP ratio compared to other large economies, while foreign invested enterprises (FIEs) and processing trade has an essential role in explaining the rapid growth in its exports. As shown in our estimation results, it is important to take into account the differences in production technology among firms with different ownership types and trade regimes.

Production for processing exports use imported intermediates more intensively than for normal exports and domestic use. Similarly, FIEs (including both wholly foreign-owned firms and joint ventures) have different patterns compared to Chinese-owned enterprises (COEs), which include state-owned, collective, and private firms, in using imported inputs. Such difference in using imported inputs exists within processing exporters as well.² Processing FIEs use imported intermediates more intensively than processing COEs.

Figure 1 shows that since the early 1990s, the share of exports by the FIEs increased to more than 50% and remained at that level, whereas the share of processing exports remained at around 50% and has started to decline since 2004. In addition, although the majority of the processing firms are foreign-owned, more than 15 percent of processing exports are conducted by local producers.

¹ Exports and GDP are measured by different accounting standards: GDP is measured in value added—a net concept, whereas export is measured in gross term containing intermediate inputs, which may cross borders many times before they become final products.

 $^{^2}$ This may indicate that multinational firms have better access to foreign components, or are more committed to import from their foreign parent firm or other affiliates within the same multinational group. We thank a referee for pointing this out.



Figure 1: Processing exports and FIEs exports in China (1995-2010)

Note: this figure gives the share of processing exports (in dark bar) and FIEs exports (in light grey bar) relative to the total value of Chinese exports, from 1995-2010. *Source*: The General Administration of Customs of China

Therefore, ignoring the differences in production technology may bias the estimates of the domestic content share in gross exports, and more so for firms that use more imported inputs. Consider, for example, the iPad, a device that is designed and owned by Apple. This device is assembled in China, and then exported to the US and other countries. In trade statistics, each iPad sold in the US adds \$275 to their trade deficit with China. However, the value added contributed by China is merely \$10, given that most parts are produced outside of China.³ Hence, although iPads accounted for \$4 billion of America's reported trade deficit with China in 2011, the *Chinese content* in the deficit was estimated to be only \$150 million.⁴ After accounting for foreign content, Johnson and Noguera (2012) showed that the controversial US-China imbalance is about 40% smaller than what was reported in official trade statistics. A precise measure of the domestic content is essential in understanding China's growing role in global trade. As discussed above, such a measure must acknowledge the pervasiveness of FIEs and processing trade.

The contributions of our paper are threefold. First, we decompose gross exports into domestic and foreign content by ownership (domestic-owned versus foreign-owned) and trade mode (processing exports versus normal exports), which results in four types of firms in each

³ Those numbers are based on a study by the Personal Computing Industry Center at University of California-Irvine, and cited by *The Economist* (2012).

⁴ Such discrepancy in trade statistics has also been reported by case studies such as Dedrick et al. (2009) for iPod, and by Xing and Detert (2011) for iPhone. Branstetter and Lardy (2008) estimate that Chinese domestic value-added accounts only for 15% of the value of exported electronic and information technology products.

sector. Separating processing firms from normal exporters to estimate the value added of Chinese exports has been investigated in literature. However, little attention has been given to the potential difference in the contribution to value added between multinational firms and local producers within processing and normal exporters. Based on the decomposition, we extend the official I/O table to incorporate the input usage difference across trade mode and ownership.⁵

Second, we developed a constrained optimization strategy to estimate these I/O coefficients as they are not readily available. Departing from existing literature which uses industry-level input-output (I/O) table, we rely on firm-level data from both industrial surveys and trade statistics to account for the heterogeneity in imported input intensity across different firm types. Thus, domestic content share in Chinese export by firm types, at both aggregate- and sector-level, are estimated. Based on the data from 2007, our results indicate that the domestic content accounted for approximately 59% of gross exports, and the rest is foreign-imported content. Furthermore, normal exports by COEs account for approximately 50% of domestic content in Chinese exports, whereas processing COEs only contributed 4.8%. Normal exports by FIEs account for approximately 17.8%, whereas processing FIEs contribute nearly 27.5%. We further break down the domestic content of exports at the industry level. We find that industries that are more sophisticated and have more foreign stakes or larger share of processing exports have lower domestic content share. Furthermore, at the sector level, the domestic content share of COEs and FIEs often diverge, even within the same trade mode. Therefore, splitting the I/O table by ownership is a necessary improvement on existing studies.

Finally, we calculate the share of domestic value-added that goes to foreign factor owners using information on the ownership structure, and further measure the distribution of income from export between foreign and domestic factor owners. Domestic value added is defined as value added that is generated within China, including value added generated by various types of enterprises operating in China. Thus, it is consistent with the GDP but is different from how the value added is distributed to different type of factor owners, which includes labor compensation and profits. The value added generated from production is converted into the income of various factor owners, which contributes to the gross national income (GNI). Based on the extended I/O structure with firm heterogeneity in each industry, we estimate that approximately 52.6% of the

⁵ We group normal exporter and non-exporter together because of their similarity in using imported inputs.

value of Chinese exports was captured by foreign factor owners (including factors owned by Taiwan, Hong Kong, and Macau).

Our study is built on a growing literature that emphasizes the vertical structure of a global value chain. Notably, Koopman, Wang, and Wei (2012, henceforth KWW) propose a formula to compute domestic and foreign contents which takes the differences in using imported inputs by processing exporters versus normal exporters into consideration. Their results are in sharp contrast to those of Hummels et al. (2001, henceforth HIY), for the latter does not recognize the role of processing trade. KWW show that the share of domestic content in Chinese manufacturing exports was about 50% before China's accession to the WTO, and has risen to more than 60% since then.

KWW's approach still depends on industry-level statistics and may have measurement bias as long as different firms within an industry have different imported input intensities. Recent studies attempt to obtain direct measures of the I/O coefficients for processing exports. For example, De La Cruz et al. (2012) show that domestic value added accounts for only 34% of Mexico's manufacturing exports on average.

Alternatively, we may rely on firm-level information to mitigate the measurement bias. This method was first attempted by Feenstra and Jensen (2012) for the US, followed by Ahmad et al. (2012) for Turkey, and Kee and Tang (2012) for Chinese processing exporters. Using a carefully cleaned sample of processing firms that merges firms' export, import, and production information from 2000 to 2006, Kee and Tang (2012) study the input choices of processing firms without resorting to the standard input–output data. They find that the average share of domestic value added in China's processing exports has risen from 35% in 2000 to 49% in 2006. This trend is very similar to that reported by KWW, which use industry level data. Similarly, Upward et al. (2012) obtain consistent results with a more general sample of Chinese firms with trade and production information from 2003 to 2006.

However, although firm-level data provide rich information, it may not be applicable to cases with less suitable data. Furthermore, relying solely on firm-level information may result in the loss of generality and prevent us from understanding the economy as a whole. Thus, our paper chooses to group firms based on their characteristics and integrate the information of firm heterogeneity with industry-level data (I/O tables) to minimize the aggregation bias in official I/O statistics. We draw information from firm-level data to make the grouping appropriate. To sum, we developed a framework and an estimation procedure that separately accounts for the production and trade activities of FIEs and COEs. Although our empirical investigation is based on Chinese data, our method is also applicable to other emerging economies that engage in massive processing trade or have significant FDI inflows, such as Mexico and Vietnam. In a general sense, correctly accounting for domestic contents in trade is essential for a precise understanding of world trade and global imbalance, as well as the distribution of gains across nations, considering that two-thirds of world trade is trade in intermediate inputs.

The rest of the paper is organized as follows: Section II provides a conceptual framework to estimate the shares of domestic value added in a country's exports. Section III presents the data and our estimation strategy. Section IV provides the results of the estimation. Section V concludes.

II. Theoretical Framework

A non-competitive I/O table as shown in Table 1 for an *n*-sector economy, is needed to estimate the domestic and foreign content of a country's exports.⁶ Z^D and Z^M are *nxn* input matrices, where superscripts denote the sources of inputs, namely, D for domestic and M for imported. Clearly imported and domestic intermediate inputs are accounted separately, capturing the poor substitutability between domestic and foreign inputs (therefore called "non-competitive").⁷ Similarly, Y^D and Y^M are *nx*1 vectors for final use. X, E, and M denote output, export, and import vectors (*nx*1), respectively. Finally, V is a 1*xn* vector for direct domestic value added (i.e., primary inputs including labor and capital).

⁶ "Domestic value added exports" is often used interchangeably in literature with "domestic content" in exports. However, the two measures are not always equal except for some special cases. As emphasized in Koopman et al. (2014), domestic content includes value added exports, as well as domestic value-added that was first exported in intermediate goods and then embodied in imports and returned home. Likewise, domestic content includes pure double counting terms, considering that intermediate goods cross borders multiple times. This is a shortcoming for all single-country I/O models, given that in such a framework, imports are assumed to be completely sourced from foreign countries. This problem may be very important for a country that exports a large amount of inputs for offshoring production, such as the US. However, this is much less of a concern for China, given that only a tiny share of China's value-added embodied in its exports returns home after further processing abroad..

⁷ Chen et al. (2004) first develop a "non-competitive" type I/O model for China. However, this model does not provide a systematic method to estimate separate input–output coefficients for producing processing exports versus those for other final demands.

	Output		Intermediate Uses	Final Use (C+I+G)	Exports	Total Output or Imports
Input		DIM	1, 2,, n	1	1	1
Intermediate	Domestic	1 n	Z^{D}	Y^{D}	E	X
Inputs	Imports	1 n	Z^{M}	Y^M	0	М
Value Added		1	V			
Total Inputs		1	X^{T}			

Table 1: the Non-Competitive I/O table

Note: Table 1 gives a typical non-competitive I/O table which distinguishes domestic inputs and imported inputs. The superscripts D and M denote domestic and imported source, respectively. Z^D and Z^M denote domestic and imported intermediate inputs, respectively; Y^D and Y^M denotes final demand vectors for domestic products and imported ones, respectively; X, E, M, and V denote outputs, exports, imports and primary inputs vector, respectively.

 $z_{ij}^{D}(z_{ij}^{M})$, an element in $Z^{D}(Z^{M})$, denotes the domestic (imported) inputs of sector *i* used by production in sector *j*. Similarly, x_{j} denotes the output of sector *j*. Furthermore, denote $A^{D} = [a_{ij}^{D}] = [z_{ij}^{D}/x_{j}]$ and $A^{M} = [a_{ij}^{M}] = [z_{ij}^{M}/x_{j}]$. By the market clearing condition, we obtain

$$X = A^D X + (Y^D + E) \tag{1}$$

$$M = A^M X + Y^M \tag{2}$$

We can express the output vector X, the total domestic content share (DVS) in final demand, and the total foreign content share (FVS) in final demand as follows.

Rearranging Equation (1), the output can be written as

$$X = (I - A^{D})^{-1}(Y^{D} + E) = B^{D}(Y^{D} + E),$$
(3)

where $B^{D} = (I - A^{D})^{-1}$ is the "Leontief inverse", which reflects both the direct and indirect usage of output in producing the final goods.

Denote $A_v = [a_{vj}] = [v_j / x_j]$ as the 1 x *n* vector of *direct domestic value added* to output ratio. Similarly, denote uA^M as the 1 x *n* vector of *imported inputs* to output ratio, where *u* is a 1 x *n* unit vector. In matrix notation, the DVS and FVS in the final demand including exports can be expressed as,

$$DVS = A_{v}(I - A^{D})^{-1} = A_{v}B^{D}$$
(4)

$$FVS = uA^{M} (I - A^{D})^{-1} = uA^{M} B^{D}$$
(5)

Thus, we can show that DVS + FVS = u because output equals total domestic content plus total foreign content for each sector.⁸ Moreover, FVS is conceptually similar to the vertical specialization share defined in HIY (2001).⁹

In a standard model, export production uses the same technology as that for domestic sales. Thus, DVS also provides the share of domestic content in exports. However, as noted by KWW (2012) and Chen et al. (2012), processing exports utilize imported inputs more intensively than do domestic sales and normal exports. Thus, relying on the standard I/O table may be biased, since it implicit assumes proportionality in using imported input. Furthermore, FIEs have played a significant role in China's economic growth, accounting for about 20% of the total output and around 50% of total export.¹⁰

Based on the observations about FIEs versus COEs as described above, we have to account for the heterogeneity of firms with different trade mode and ownership to correctly estimate domestic content. Specifically, we split firms into four groups, namely, processing exports by COEs (CP), processing exports by FIEs (FP), normal exports and domestic sales by COEs (CN), and normal exports and domestic sales by FIEs (FN).¹¹ In Section 3, we describe the comparison of import pattern across these four groups and rationalize our current way of splitting firms.

We extend the standard I/O table to incorporate such splits, as shown in Table 2.

For input-output matrices Z, the first letter in the superscript denotes the ownership of the input supplier; the second letter denotes the ownership of the user; and the last letter denotes the export mode. C denotes COEs, F denotes FIEs, M denotes imports, P denotes processing, and N denotes normal exports and domestic sales. For example, an element in Z^{CFP} , namely z_{ij}^{CFP} , denotes the intermediate inputs produced by COEs in sector *i* and used by FIEs in sector *j* for processing exports. For the other vectors, the first letter in the superscript denotes ownership and the second denotes export mode. For example, E^{FN} denotes the normal exports of FIEs. If we

⁸ $DVS + FVS = (A_v + uA^M)(I - A^D)^{-1} = (u - uA^D)(I - A^D)^{-1} = u$.

⁹ HIY (2001) define the measure of vertical specialization as the imported goods that are used as inputs to produce a country's export goods, assuming the use intensity of imported inputs in the same between production for exports and production for domestic sales.

¹⁰ A large proportion of processing trade is conducted by FIEs. However, although the share of processing exports remained steady at more than 50% during the last decade, FIE's share in China's total exports increases steadily from 31% in 1995 to 58% in 2006, and slightly decreases afterwards (see Figure 1).

¹¹ We assume that normal exports employ inputs in the same proportion as production for domestic sales, following KWW (2012). We have conducted further statistical tests based on detailed firm-level data to determine the best split of firm types at sector level to minimize aggregation errors caused by firm heterogeneity. A detailed discussion of the split and the comparison of key variables between groups are provided in the succeeding section.

ignore the difference in input usage between foreign and domestic firms within a trade mode (i.e., processing versus normal), Table 2 collapses to the I/O table proposed by KWW. It can be shown that HIY's results are a special case by KWW, in which no difference is found between the use of imported inputs in processing and normal exports.

Output		Intermedia CO	te use by Es	Intermediat FIE	te use by s	Final	Exports	Gross Output or
Input		N	Р	N	Р	use		Imports
Domestic Intermediate	Ν	Z^{CCN}	Z^{CCP}	Z^{CFN}	Z^{CFP}	Y ^C	E^{CN}	X^{CN}
input of COEs	Р	0	0	0	0	0	E^{CP}	X^{CP}
Domestic Intermediate	Ν	Z^{FCN}	Z^{FCP}	Z ^{FFN}	Z^{FFP}	Y ^F	E^{FN}	X ^{FN}
input of FIEs	Р	0	0	0	0	0	E^{FP}	X^{FP}
Imports		Z^{MCN}	Z^{MCP}	Z^{MFN}	$Z^{\rm MFP}$	Y^M	0	М
Value added		V^{CN}	V^{CP}	V^{FN}	V^{FP}			
Gross Input		$\left(X^{CN}\right)^{T}$	$\left(X^{CP}\right)^T$	$\left(X^{FN} ight)^T$	$\left(X^{FP}\right)^{T}$			
Foreign income	e	$G^{^{CN}}$	$G^{\scriptscriptstyle CP}$	$G^{^{F\!N}}$	$G^{{\scriptscriptstyle F\!P}}$			

Table 2: I/O table by ownership and trade mode

Note: Table 2 gives the expanded I/O table that distinguishes domestic inputs by ownership and input usages by ownership and trade mode. Superscript C and F represent COEs and FIEs, respectively; M denotes imports; P and N represent processing exports, domestic sales and normal exports, respectively. X is gross output, E is exports, M is imports, Z denotes intermediate input matrices, Y is total final demand except for exports, V is value added, and G is foreign income. Superscript T represents transpose.

Furthermore, we can express the input coefficients as n x n matrices:

$$A^{olk} = [a_{ij}^{olk}] = [z_{ij}^{olk} / x_j^{lk}]$$
 where $o = C, F, M; \quad l = C, F; \quad k = N, P,$

The direct value added coefficients as 1 x n vectors can be expressed as

 $A_{v}^{lk} = [a_{vj}^{lk}] = [v_{j}^{lk} / x_{j}^{lk}]$

Similar to the derivation of Equation (3), we can write the extended Leontief inverse as^{12}

¹² A complete expression for Equation (6) is shown in the online Appendix A. The whole appendix could be downloaded from <u>http://mahong.weebly.com/uploads/2/7/0/9/27093249/201312</u> online appendix resubmission.pdf.

$$\overline{B} = \begin{bmatrix} B^{CCN} & B^{CCP} & B^{CFN} & B^{CFP} \\ 0 & I & 0 & 0 \\ B^{FCN} & B^{FCP} & B^{FFN} & B^{FFP} \\ 0 & 0 & 0 & I \end{bmatrix} = (I - \overline{A^{D}})^{-1} = \begin{bmatrix} I - A^{CCN} & -A^{CCP} & -A^{CFN} & -A^{CFP} \\ 0 & I & 0 & 0 \\ -A^{FCN} & -A^{FCP} & I - A^{FFN} & -A^{FFP} \\ 0 & 0 & 0 & I \end{bmatrix}^{-1}$$
(6)

Analogous to Equation (4), the total domestic content share is,

$$\overline{DVS} \equiv \begin{pmatrix} DVS^{CN} & DVS^{CP} & DVS^{FN} & DVS^{FP} \end{pmatrix} = \overline{A_{\nu}B},$$
(7)

where $\overline{A_{\nu}} = \begin{pmatrix} A_{\nu}^{CN} & A_{\nu}^{CP} & A_{\nu}^{FN} & A_{\nu}^{FP} \end{pmatrix}$ and \overline{B} is given in (6). Thus for each sector, we will obtain domestic content share for firms belonging to each of the four types. The weighted sum gives the sector-level total DVS as

$$TDVS = \sum_{l} \sum_{k} DVS^{lk} \hat{s}^{lk} \qquad \text{where} \quad l = C, F; \quad k = N, P \tag{8}$$

where each \hat{s} is an *n* x *n* diagonal matrix, with the export share of each type firm as the diagonal term.

Similar to Equation (5), the total foreign contents share is

$$\overline{FVS} = \left(FVS^{CN} \quad FVS^{CP} \quad FVS^{FN} \quad FVS^{FP}\right) = u\overline{A^{M}}\overline{B}$$
(9)

where $\overline{A^{M}} = \begin{pmatrix} A^{MCN} & A^{MCP} & A^{MFN} & A^{MFP} \end{pmatrix}$. The total FVS is

$$TFVS = \sum_{l} \sum_{k} FVS^{lk} \hat{s}^{lk} = u - TDVS$$
⁽¹⁰⁾

Furthermore, we augment the extended I/O table with a row that represents the income that goes to foreign factor owners, and we define foreign income share (FIS) as

$$A_G^{lk} = [a_{Gj}^{lk}] = [g_j^{lk} / x_j^{lk}],$$
(11)

where g_j^{lk} denotes foreign income for type lk firms. FIS differs from foreign content share for two reasons: first, a proportion of the profits of FIEs may go to its foreign investor; second, wage compensation to foreign workers working in FIEs and COEs. FIS in domestic content can be computed as:

$$\overline{FIS} \equiv \left(FIS^{CN} \quad FIS^{CP} \quad FIS^{FN} \quad FIS^{FP}\right) = \overline{A_G}\overline{B}$$
(12)

where $\overline{A_G} = \begin{pmatrix} A_G^{CN} & A_G^{CP} & A_G^{FN} & A_G^{FP} \end{pmatrix}$. As constructed, FIS represents the domestic content that serves as the income earned by foreign factor owners. Thus, the total domestic national income share in gross exports can be expressed as a weighted sum of the domestic content after eliminating the portion of the content that belongs to foreign factor owner:

$$TDNIS = \sum_{l} \sum_{k} (DVS^{lk} - FIS^{lk}) \hat{s}^{lk} \quad \text{where} \quad l = C, F; \quad k = N, P$$
(13)

Correspondingly, the total foreign income share of gross exports is

$$TFNIS = \sum_{k} \sum_{k} (FVS^{lk} + FIS^{lk}) \hat{s}^{lk}$$
(14)

III. Data and Estimation Strategy

III.A: Data construction

Our data come from several sources. First, we obtain the official I/O table for the benchmark year 2007 with 135 sectors (including 80 manufacturing sectors), published by the National Bureau of Statistics of China (NBSC). This is a standard I/O table with a similar structure as Table 1. However, it does *not* separately account for domestic and imported inputs, and thus, the imported inputs are used across sectors in proportion to the domestic inputs. With this I/O table, we obtain the intermediate input matrices (Z), the value added vector (V), the output vector (X), the exports vector (E), the imports vector (M), and a final demand (excluding exports) vector (Y). To further extend the table to account for how different types of firms use imported intermediate input in their production and how much they export, we need more detailed import and export data based on ownership and trade mode.

Such data come from combining two large firm-level datasets. The first is the Annual Surveys of Industrial Production (ASIP) compiled by the NBSC, which includes firm-level information on balance sheet, production, ownership, etc. It covers all state-owned enterprises (SOEs) and all non-state enterprises with total sales exceeding 5 million RMB in the industrial sector (including mining, manufacturing, and public utilities) from 1998 to 2007 (Brandt, 2012). The second dataset is the firm-level export and import data for 2007, from China's General Administration of Customs (CGAC). It provides information on firm-specific import of intermediate inputs and total export. Intermediate inputs are identified using the UN Broad Economic Classification (BEC). By matching firms from the two datasets, ¹³ we can categorize industrial firms based on ownership and trade mode, namely, processing exports by COEs (CP),

¹³ We first eliminated observations with unreasonable values and outliers (i.e., observations with negative gross output, or negative input value, zero employment, and the top and bottom 0.5 percentiles for key variables). As a result, we obtain a sample with 301,774 firms, 92,628 of which are exporters. Not surprisingly, many exporters are also importing: 38,025 firms are exporting and importing at the same time. Furthermore, processing exporters are major contributors to Chinese exports; altogether, a total of 26,611 exporters are engaged in processing trade. More than one-fifth of the firms are foreign-owned firms or joint ventures. The details of the datasets, as well as the detailed matching procedure, are presented in the online Appendix B.

processing exports by FIEs (FP), normal exports or domestic sales by COEs (CN); normal exports or domestic sales by FIEs (FN). Most non-exporters are COEs, whereas most processing exporters are FIEs. Table 3 summarizes the mean, the variance, and different percentiles for imported input intensity, which are divided total input or total output, across four types of exporters.¹⁴ On average, foreign owned processing exporters (FP firms) have the highest imported input intensity (0.436), whereas Chinese owned normal exporters (CN firms) have the lowest (0.003). In terms of variance, FP firms have the highest variance as well, whereas CN firms have the lowest. Within processing exporters, FP firms have higher imported input intensity than CP firms for all percentiles. CP firms have higher imported input intensity than CN and FN firms.

Туре	Freq.	mean	variance	p25	p50	p75	p90	p99
Panel A: Impo	rted input o	over input						
СР	4,112	0.144	0.127	0.003	0.028	0.124	0.392	1.635
FP	22,495	0.436	0.655	0.049	0.217	0.536	0.899	4.300
CN	41,885	0.003	0.000	0.000	0.000	0.000	0.000	0.086
FN	24,136	0.037	0.015	0.000	0.000	0.002	0.089	0.687
Total		0.123	0.201	0.000	0.000	0.041	0.372	1.612
Panel B: Impo	rted input o	over output						
СР	4,112	0.103	0.059	0.002	0.021	0.096	0.288	1.035
FP	22,495	0.287	0.202	0.036	0.159	0.388	0.656	2.113
CN	41,885	0.002	0.000	0.000	0.000	0.000	0.000	0.063
FN	24,136	0.027	0.009	0.000	0.000	0.001	0.063	0.494
Total		0.082	0.068	0.000	0.000	0.030	0.270	0.969

Table 3: Mean and percentile comparisons in input intensity among groups

Note: Table 3 presents the mean and percentile comparisons in input intensity among groups by ownership and trade mode. The top panel provides summary statistics for the share of imported inputs over total inputs. The bottom panel provides summary statistics for the share of imported input over total output. *Source*: the merged sample of customs data and ASIP

Using the Scheffe multiple-comparison tests, Table 4 provides simple cross-table mean comparison for the imported input intensity across different groups. All group means are significantly different from each other. In particular, we stress that (1) within processing exports, FP firms, on average, use significantly more imported input over total input or output as compared with CP firms; (2) within FIEs, FN firms, on average, use a significantly less

¹⁴ We lack information on imported input intensity for non-exporters. Non-exporters, as well as normal exporters, however, may use imported inputs indirectly by sourcing the inputs from direct importers or intermediaries. Therefore, our estimates of import intensity for normal exporters may be understated.

proportion of imported input than FP firms; (3) within COEs, CN firms, on average, use significantly less proportion of imported input than CP firms. Firms under different groups have distinct patterns of input usage. Similar patterns are found when the mean and variance of import intensity are compared across four types of firms within each manufacturing sector.¹⁵

Comparison of	Imported	input / inpu	ıt	Imported input / output			
Row Mean - Column Mean	СР	FP	CN	СР	FP	CN	
FP	0.291			0.184			
CN	-0.141	-0.433		-0.101	-0.285		
FN	-0.107	-0.399	0.034	-0.076	-0.259	0.025	

Table 4: Mean differences in imported input intensity, Scheffe multiple-comparison test

Note: Table 4 does the analysis of variance using the Scheffe multiple-comparison test. Input intensity is defined as the value of imported input over total output. Differences defined as row mean - column mean. All are significant at 1% level. *Source*: the merged sample of customs data and ASIP

Moreover, splitting normal firms further into normal exporters versus non-exporters (i.e., firms that sell only at domestic market) may be desirable, as this may further reduce the aggregate bias. However, we lack information on imported input intensity for non-exporters. Although there are a large number of non-exporting producers in the ASIP data, no information is available on where their intermediate inputs are sourced from. More importantly, under the processing regime, firms can obtain imported intermediate inputs free of tariffs and value-added taxes (VAT).¹⁶ Therefore, when importing inputs is necessary to fulfill its export contacts, a firm tends to import under processing regime. On the other hand, the difference between normal exporters and firms that sell only to domestic market in terms of imported input use intensity may not be as significant to justify further separation as compared with other countries without such incentives. On the basis of these two reasons, we decide not to separate normal exporters and domestic non-exporters in our current study.

Furthermore, we are interested in knowing the income transfer from domestic valueadded to foreign factor owners as well. Such information by sector is available from the balance of payment (BOP) table compiled by the People's Bank of China (PBC). This includes

¹⁵ The results are presented in online Appendix C (Table C1).

¹⁶ VAT ranges from 13-17 percent. In addition, being recognized as "processing importer" reduces the procedures for tax "collection" and then "rebate", since value-added tax is not collected for imported intermediate goods used to produce exports.

investment income and employee compensation at the sector level.¹⁷ To split factor income for each type of firms within a sector, we use the combined dataset described above, which also provides information for all four types of industrial firms on foreign share in total paid-in capital, and detail income decomposition in value added, including labor compensation and capital income (i.e., operation surplus plus depreciation). Thus, we can pin down the foreign share in domestic value-added for industrial sectors. We use foreign share in registered capital as reported in the 2008 China Statistical Yearbook for primary and tertiary sectors to obtain the initial value of the foreign capital income share for FP and FN firms. Assuming that a foreign employee only works for foreign firms, we split compensation to foreign employees in the sector level income into the FP and FN's foreign employee compensation based on the proportion of total employee compensation for FP and FN.

III.B. Estimation Strategy

To extend the official I/O table to account for different input usage across ownership and trade mode, we augmented the official I/O table with information from other available data sources presented above. Likewise, we split the official I/O table from NBSC into blocks by ownership and trade mode. This process ensures that all the aggregate numbers and the balance conditions in the official I/O table are met by the estimated new I/O table with firm heterogeneity.

The key variables in our estimation model are listed as follows, including both variables to be estimated and those that we have obtained from the data through construction.

Variables	Definition	#
(o, <i>l</i> = <i>C</i> , <i>F</i> , <i>k</i> = <i>N</i> , <i>P</i>)		unknowns
	variables to be estimated	
7^{olk}	Intermediate good <i>i</i> produced by type <i>o</i> firms and used by type <i>l</i> firms and trade	$8n^2$
z. _{ij}	mode k in sector j	
7^{mlk}	Intermediate good <i>i</i> imported to be used by firms of type <i>l</i> in sector j for trade	$4n^2$
∠ _{ij}	mode k	
v_{k}^{lk}	Value added by <i>l type</i> firms in sector <i>j</i> for trade mode <i>k</i>	4n
v j		
v^l .	Final goods used domestically produced by <i>l</i> type firms in sector <i>j</i>	2n
5]		
	variables with available data	
X. V. V. P. M.	Output, value added, final demand, and export and imported intermediates by	
x_j, v_j, y_j, c_j, m_j	sector, from the official I/O table	

¹⁷ More specifically, we use the sector-level income table (expenditure side) in BOP. Foreign investment income accounts for nearly 95% of the total foreign factor income.

Z _{ij}	Goods i used as intermediate inputs in sector j , without distinguishing ownership and trade mode, from the official I/O table	
x_j^l, v_j^l	Output and value added by <i>type l</i> firms in sector <i>j</i> , known from the ASIP data the official I/O table, and China Statistical Yearbook.	
y_j^m	Imported final goods of sector <i>j</i> , known from trade statistics and the official I/O table.	
m_j^{lk}	Normal and processing imported intermediate inputs of sector j by FIEs and COEs respectively, known from the combined dataset and the official I/O table	
e_{j}^{lk}	Normal and processing exports of sector j by FIEs and COEs respectively, known from the combined dataset and the official I/O table	

We now obtain the data for output, export, value added, imports for final demand, and imported intermediate inputs, by type and sector. These data will be used as controls for our estimations. To be more specific, sector-level gross output and direct value added by each firm type are obtained from the ASIP data and the China Statistical Yearbook.¹⁸ For example, x_j can be divided into total output of COEs (x_j^c) and FIEs (x_j^f). $x_j^{cp} = e_j^{cp}$ denotes the processing exports of COEs. $x_j^{fp} = e_j^{fp}$ denotes processing exports of FIEs. The output of COEs for normal exports and domestic use, x_j^{cn} , denotes the difference between output and processing exports ($x_j^c - e_j^{cp}$). Similarly, the output of FIEs for normal exports and domestic use x_j^{fn} is $x_j^f - e_j^{fp}$. Using the same approach, we can also separate value added v_j into v_j^c , v_j^f .

Our estimation procedure is performed using a quadratic programming model. As indicated above, our method involves estimating eight inter-industry-between-firm-type domestic input transaction matrix Z^{olk} , four inter-industry-between-firm-type imported input transaction matrix Z^{mlk} , four sector-level value added vectors V^{lk} , and two domestic final demand vectors Y^o . Assuming there are *n* sectors, our estimation will involve $12n^2$ unknowns for intermediate inputs, 4n unknowns for value added, and 2n unknowns for the final demand. First, we make conjectures about their values based on the trade statistics, the ASIP, and the official benchmark I/O table. These conjectured values are then used in our estimation as initial values.

For domestic final demand, we set the initial value $y0_i^o$ as residual:

$$y0_{i}^{o} = \left[(x_{i}^{o} - e_{i}^{o}) / (x_{i} - e_{i}) \right] (y_{i} - y_{i}^{m}),$$
(15)

¹⁸ ASIP dataset only includes industrial firms. For other industries, such as construction and transportation, we can obtain the sector-level value-added data from the Statistical Yearbooks. The shares of FIEs are estimated based on total sales of FIEs or total registered capital by FIEs.

where $(x_i^o - e_i^o)/(x_i - e_i)$ indicates the fraction of domestic use (exclusive of export) by *o* type firms out of total domestic use in sector *i*.

The initial values of value added, $v0_j^{lk}$, are from decomposing v_j^c , v_j^f based on the combined data from the ASIP and the Customs statistics.

The initial values of intermediate imports use coefficients that are generated by allocating sector *i*'s imported intermediates m_i^{lk} in proportion to input *i*'s usage in sector *j*

$$z O_{ij}^{mlk} = \left[z_{ij} / \sum_{j=1}^{n} z_{ij} \right] m_i^{lk} , \qquad (16)$$

where m_i^{lk} can be estimated from detailed trade statistics, whereas z_{ij} are from the official I/O table. The fraction in brackets denotes the proportion of input in sector *i* that are used in sector *j*.

The initial values for domestically produced intermediates are generated through two steps. First, we compute for the total domestic product i used as intermediate inputs in sector j as a residual of the total intermediate inputs and imported intermediate inputs:

$$z_{ij}^{d} = z_{ij} - \sum_{l} \sum_{k} z 0_{ij}^{mlk}$$
(17)

Second, we assume a proportional usage of z_{ij}^d :

$$z O_{ij}^{olk} = \left[(x_i^o - e_i^o) / (x_i - e_i) \right] \left[x_j^{lk} / x_j \right] z_{ij}^d$$
(18)

where $\left[x_{j}^{lk}/x_{j}\right]$ denotes the fraction of output of type *lk* firms in sector j, with *o*,*l*=*C*,*F*, *k*=*N*,*P*.

Nevertheless, these initial conjectures are not guaranteed to satisfy various economic and statistical restrictions. Therefore, we cast the estimation problem as a constrained optimization procedure. In particular, the optimization program is specified to minimize the following objective function:

$$\operatorname{Min} S = \sum_{i=1}^{n} \sum_{j=1}^{n} \sum_{o} \sum_{l} \sum_{k} \left[(z_{ij}^{olk} - z \mathbf{0}_{ij}^{olk})^{2} / z \mathbf{0}_{ij}^{olk} \right]$$

$$+ \sum_{i=1}^{n} \sum_{j=1}^{n} \sum_{l} \sum_{k} \sum_{k} \left[(z_{ij}^{mlk} - z \mathbf{0}_{ij}^{mlk})^{2} / z \mathbf{0}_{ij}^{mlk} \right]$$

$$+ \sum_{j=1}^{n} \sum_{l} \sum_{k} \sum_{k} \left[(v_{j}^{lk} - v \mathbf{0}_{j}^{lk})^{2} / v \mathbf{0}_{j}^{lk} \right]$$

$$+ \sum_{i=1}^{n} \sum_{o} \left[(y_{i}^{o} - y \mathbf{0}_{i}^{o})^{2} / y \mathbf{0}_{i}^{o} \right]$$

$$(19)$$

where z's, v's and y's are the variables to be estimated, whereas z0's, v0's and y0's are the initial values that have been specified above. Equation (19) minimizes the sum of the differences of the estimated values from their initial values. Furthermore, the results of this minimization problem should satisfy the balance conditions of both the official I/O table and the extended table, in addition to other regularities that we specify below.

Constraint set 1: row sum identities required by the I/O table

$$\sum_{j=1}^{n} \sum_{k} z_{ij}^{olk} + y_{i}^{o} = x_{i}^{o} - e_{i}^{o}$$
(20)

$$\sum_{j=1}^{n} \sum_{l} \sum_{k} z_{ij}^{mlk} + y_{i}^{m} = m_{i}$$
(21)

Constraint set 2: column sum identities required by the I/O table

$$\sum_{i=1}^{n} \sum_{o} (z_{ij}^{olk} + z_{ij}^{mlk}) + v_{j}^{lk} = x_{j}^{lk}$$
(22)

Constraint set 3: adding up conditions for intermediate inputs

$$\sum_{o} \sum_{l} \sum_{k} z_{ij}^{olk} + \sum_{l} \sum_{k} z_{ij}^{mlk} = z_{ij}$$
(23)

Constraint set 4: adding up conditions for import intermediate inputs

$$\sum_{j=1}^{n} z_{ij}^{mlk} = m_i^{lk}$$
(24)

Constraint set 5: adding up conditions for value added and final use

$$\sum_{l} \sum_{k} v_{j}^{lk} = v_{j} \tag{25}$$

$$\sum_{o} y_i^o + y_i^m = y_i \tag{26}$$

Constraint set 6: non-negativity constraints

$$z_{ij}^{olk}, \quad z_{ij}^{mlk}, \quad v_j^{lk}, \quad y^m \ge 0$$
 (27)

The economic meanings of the six sets of constraints are straightforward. Equations (20)-(21) pertain to the row sum identities for the expanded I/O account, which is also the market clearing condition. These equations state that the total gross output of sector i should be equal to the sum of final demand and exports, plus domestic use as intermediates by all four firm types across all sectors. Similarly, the total imports should be equal to the imported intermediate inputs used across firm types and sectors, plus imports delivered to final users. Equation (22) provides the column sum identities, maintaining that the total gross inputs of sector j should be equal the intermediate inputs, supplied by four firm types across sectors and imported intermediate inputs,

plus primary inputs. Equations (23) to (26) are a set of adding-up constraints that ensure the consistency of the solution from the model with official statistics on sector-level trade and transactions within the industry. Finally, Equation (27) indicates the non-negativity condition.¹⁹

IV. Results

Using the estimation method described in the previous section, we estimated an extended non-competitive I/O table with separate production accounts by firm ownership and trade mode.²⁰ The results of domestic and foreign content share of exports by the four type firms in each industry are estimated based on this estimated I/O table.

IV. A. Share of domestic and foreign content

Table 5 presents our estimation results for the DVS and FVS shares in China's processing and normal exports by COEs and FIEs from 2007. These data were calculated using Equations (7) to (10). Columns (1) to (4) report the direct DVS, total DVS, direct FVS, and total FVS, respectively. The direct DVS and FVS are the direct value added incurred in each sector as income to primary inputs, whereas the total DVS and FVS account for the iterated usage of inputs from other sectors. Several observations have to be stressed, particularly regarding the DVS in columns (1) and (2). First, comparing rows (1) and (4), although the direct DVS for COEs is very close to that for FIEs (20.92 versus 18.12), the COEs' total DVS is larger (75.07 versus 47.13). Second, within each ownership type, the difference between normal and processing exports in total DVS is larger than the difference in direct DVS. As expected, normal exports have a higher domestic content share as compared with processing exports, regardless of the ownership type. Furthermore, COEs have slightly higher total DVS than FIEs for normal exports, whereas they have slightly lower total DVS than FIEs for processing exports. Third, the total DVS of COEs within the same trade mode is very close to that of FIEs. For example, if we compare rows (2) and (5) with processing exporters, the total DVS for COEs is 35.46, whereas that for FIEs is 37.3. Thus, one may conclude that the difference of total DVS between COEs

¹⁹ The partition among the five parts of imports $(m_i^{cp}, m_i^{cn}, m_i^{fp}, m_i^{fn} \text{ and } y_i^m)$ based on Custom import statistics and UN BEC classification allows for the slight adjustment only to the extent that a feasible solution from model could be obtained. This reconciliation procedure is implemented in GAMS (Brooke et al, 2005). Related computer programs and data files will be available at the USITC website for download.

²⁰ In the online Appendix C, we provide all trade share parameters that we use in estimation (Table C2).

and FIEs is mainly driven by the ratio of processing to non-processing exports. However, DVS differs substantially across ownership in certain sectors, as shown in Table 6.

Total	Merchandise	Direct DVS [1]	Total DVS [2]	Total FVS [3]
[1] Total exports by COEs		20.92	75.07	24.92
[2]	Processing exports by COEs	15.58	35.46	64.52
[3]	Normal exports by COEs	22.14	84.11	15.89
[4] <i>To</i>	tal exports by FIEs	18.12	47.13	52.86
[5]	Processing exports by FIEs	16.64	37.3	62.7
[6]	Normal exports by FIEs	23	79.53	20.46
[7] To	tal gross exports	19.33	59.17	40.82

 Table 5: The domestic and foreign content share of China's exports (%), 2007

Note: Table 5 presents our estimation results for the share of domestic (DVS) and foreign (FVS) content in China's processing and normal exports by COEs and FIEs in 2007.

Overall, about 59.2% of Chinese exports are domestic content, whereas 40.8% are foreign content, as shown in row (7) of Table 5. In 2007, the gross value of Chinese exports is 1.22 trillion US dollars (USD), and thus the gross domestic content in export is 720 billion USD. Table 6 shows the composition of the gross exports and gross domestic content in exports by firm type. Almost half of the domestic content is attributed to COEs' normal exports, and 4.8% of domestic content is attributed to the processing exports of COEs, and thus COEs contribute nearly 55% of gross domestic content in exports. The rest of the content can be attributed to the FIEs; in particular normal FIEs exports account for about 17.8%, whereas processing FIEs contribute nearly 27.5%.

		COEs exports		FIEs exports				
	normal	processing	total	normal	processing	total		
% share in gross exports	35.09	8.00	43.09	13.25	43.66	56.91		
% share in total domestic content of exports	49.87	4.80	54.67	17.82	27.52	45.34		

Table 6: Distribution of Domestic Content by Firm Type (%), 2007

Note: Table 6 shows the decomposition of gross exports and gross domestic content in exports, by firm ownership and trade mode. *Source*: China trade statistics and authors' own calculation

IV. B. Domestic and foreign content share by sectors

Our findings on the heterogeneity in domestic content share across firm types are consistent with those of KWW (2012) and Chen et al. (2012).²¹ In particular, KWW estimate the total DVS is 83.9% for normal exports, and 38.7% for processing exports. Overall, they find that the total DVS for Chinese exports is 60.6%. In aggregate and by trade mode, our DVS estimates are very close to theirs. However, we demonstrate that across sectors, the difference in DVS by trade mode varies within ownership, which was not captured by their studies.

In Table 7, we present the DVS by firm types for manufacturing sectors.²² Columns (1) and (2) show the DVS for COEs and FIES of processing exports, respectively, whereas columns (3) to (4) show those for two types of normal exports. Column (5) lists the total DVS for the total exports, which is a weighted average of the DVS of four types. Sorted by total DVS in an ascending order, 11 sectors have domestic content share over 80 percent. These sectors with relatively high share of domestic content include some labor-intensive sectors, such as textile, porcelain, cement, and plaster sectors, as well as some capital- and resource-intensive sectors, such as steelmaking and coking. These sectors usually have very low processing share [column (9)]. By contrast, most high-tech or sophisticated industries have a low share of domestic content. In addition, 10 sectors have domestic content share less than 50 percent, including electronic components, telecommunication equipment, and computers. Likewise, those sectors have high share of processing exports and high share of FIEs.

²¹ All three studies present much lower DVS than HIY (2001), which assume that the intensity in the use of imported inputs is the same between production for exports and production for domestic sales. This finding indicates that ignoring the heterogeneity across types of firms in their production technology will overestimate domestic content share.

²² To facilitate comparison, we focus on our sample of 57 manufacturing sectors, which were also reported in KWW.

Ta	able 7	7: Decon	position	of Domes	tic Content	bv	Sectors	(%)	, 2007
						· · · ·		· · · /	,

	MWZ							processing	
Industry	СР	FP	CN	FN	Total	Р	N	Total	share
	[1]	[2]	[3]	[4]	[5]	[6]	[7]	[8]	[9]
Household Audiovisual Apparatus	27.5	25.4	77.3	74.7	30.3	29.6	75.9	32.6	91.0
Electronic Component	36.8	21.9	72.8	76.7	31.4	23.1	77.5	32.3	83.1
Cultural and office equipment	37.7	27.6	75.2	75.5	33.1	33.1	74.1	36.5	90.5
Telecommunication equipment	33.3	33.2	74.7	66.8	39.6	35.3	75.2	43.6	83.1
Generators	31.3	27.0	80.5	71.1	40.4	51.2	80.3	66.6	73.9
Electronic computer	19.7	42.4	66.2	70.9	42.5	33.0	75.7	33.9	94.7
Measuring Instruments	41.5	34.6	81.0	71.9	43.2	37.8	80.0	45.8	81.3
Petroleum feline and Nuclear Fuel	15.2	37.5	68.1	69.2	45.5	20.1	68.7	44.4	50.6
Household electric appliances	26.8	27.2	82.3	78.2	46.4	35.6	82.0	51.8	64.1
Synthetic Materials	40.3	34.2	75.3	69.4	47.6	34.0	76.4	47.7	67.7
Ship building	39.7	65.2	84.8	74.7	50.1	39.1	83.9	43.8	86.7
Plastic	31.1	27.6	79.3	75.7	52.5	31.1	80.8	55.1	51.7
Rubber	25.2	31.1	81.5	77.5	53.8	27.0	81.8	53.4	51.8
Articles for Culture, Education and Sports Activities	34.3	46.4	83.2	74.4	56.4	45.6	83.0	58.4	66.2
Other special industrial equipment	46.9	39.1	82.9	74.6	57.7	43.0	82.5	65.2	55.9
Chemical Fiber	38.0	51.0	76.2	74.8	59.7	51.9	76.4	62.6	54.4
Other electric machinery and equipment	37.8	36.9	79.6	76.9	61.5	33.7	80.3	52.1	41.2
Other electronic and communication equipment	61.8	60.8	81.2	72.4	64.4	34.7	68.0	39.7	80.6
Paper and Paper Products	59.0	50.8	85.0	82.0	64.6	57.6	85.5	69.2	60.8
Metal products	39.1	30.5	85.0	82.3	66.9	39.7	85.1	/0.1	32.7
Special Chemical Products	62.5	51.8	76.5	73.3	68.5	34.0	76.7	61.6	31.2
Nonferrous metal pressing	41.1	53.0	77.7	77.5	69.0 5 0 0	56.1	78.6	/1.2	30.5
Glass and Its Products	63.5	51.0	84.5	78.0	70.8	59.0	83.3	/6./	38.4
Leather, fur, down and related products	35.3	41.1	89.3	89.1	71.1	40.4	90.4	69.2	36.7
Nonferrous metal smelting	49.3	57.3	/6./	63.7	71.4	56.4	76.2	73.3	18.6
Furniture	53.0	42.2	8/.1	84.7	71.6	56.1	86.7	76.2	34.2
Other transport equipment	57.0	50.2	82.3	/3.8	71.6	54.9	81.0	/3.8	28.0
Railroad transport equipment	58.7	62.1	81.1	67.3	71.9	54.1	11.1	69.0	37.0
Other manufacturing products	39.8	53.5	80.0	84.4	72.0	48.1	86.5	72.3	36.3
Pesticides Motor visibles	59.8 42.0	00.2 45.2	/ 5.0	75.9	13.2	55.0 47.4	/3.9	72.9	5.4 22.5
Motor venicies	43.0	45.5	83.4	79.0	73.5	47.4	84.0	75.5	23.5
Basic Chemical Raw Materials	28.0	54.5 19 7	81.5	/2.4	13.5	42.5	80.8	74.9	10.0
Wearing apparel	26.0	40.7	04.7 20.7	80.0 88 0	13.5	52.0	83.0 80.5	75.0	20.0
Rolling of Stool	22.0	59.5 52.9	89.7 80.4	00.2 70.2	74.5	52.9	89.5	79.0 77 8	29.1
Rolling of Steel Boiler, angings and turking	50.1	52.8 62.7	80.4 82.2	70.2	15.2	32.9 20 7	80.0 81.6	70.6	0.5 25 0
Iron smalting	39.1 40.5	45.5	02.3 76.9	72.0	15.5 75 7	50.7	81.0 75.0	70.0	23.0
A original type, forestry, and fishing mashingry	40.5	43.5	/0.8	73.9	15.1	50.0	13.9	75.0	1.1
Chamical Broducts for Daily Usa	64.8	66.8	82.0 70.0	73.0 91.5	76.2	591	80.0	73.0	22.1
Cotton tavtilas	20.0	12.2	79.9 87.0	81.3 86.0	70.5	JO.4	80.8 88 0	78.0	30.0
Smalting of Ferroallov	50.0 62.4	43.3 57.1	01.9 78 0	71.5	70.7	43.0 53.3	88.0 75 7	75.6	21.0
Woolen textiles	61.0	64.0	00.4	70.3	77.5	57.0	80.4	76.0	30.8
Fortilizora	55.7	60.0	90.4	79.5	79.2	57.9	09.4 01.0	70.9	12.0
Metalworking machinery	55.7 63.4	61.6	02.0 82.8	78.2	70.5	56.8	81.0	77.3	15.0
Dainte Drinting Inke Digments and Similar Droducts	63.0	68.8	86.0	85.5	70.4	56.8	76.5	72.6	30.0
Printing Reproduction of Recording Media	63.0	68.8	86.9	85.5	79.0	61.0	86.4	76.5	39.0
Wood Bamboo Battan Palm and Straw Products	57.4	61.3	84.8	82.1	80.2	58.4	84.6	80.4	16.1
Pottery and Porcelain	57.4 65.0	63.8	84.3	78.2	81 1	58 2	83 /	82.0	8.4
Steelmaking	N/A	17 /	817	76 0	81 2	51.2	80.8	80.8	0.4
Textiles productions	45.2	60.1	88.8	86.5	81.2	54.9	88.4	82.4	20.7
Knitted and crocheted fabrics and articles	40.8	<u>40</u> 3	87.9	87.0	81.5	51.6	88 2	82.5	15.6
Other non-metallic mineral products		62 7	87.5	727	82 5	56.6	86.0	83.0	11.3
Fire-resistant Materials	63.1	60.1	88.7	813	82.5	55.0	86.6	84 7	9.4
Medicines	60.0	58.1	87.9	83.7	82.0	37.5	87.6	803	13.9
Hemn textiles	64.1	62.5	88.0	80.7	84.8	56.8	86.6	83.9	9.9
Cement Lime and Plaster	61 2	43.0	89.8	81 7	86 2	52.9	89.0	88.4	28
Coking	N/A	-3.0 N/Δ	90.8	73.9	88.8	N/A	89.6	89.6	0.0
Total Merchandise	35.5	37.3	84.1	79.5	59.2	37.3	84.0	60.6	50.7

Note: This table gives the decomposition of domestic content by manufacturing sectors. Columns [1]-[5] present DVS estimated in this paper, while columns [6]-[8] present DVS estimates by KWW. Column [9] gives the share of processing exports in total exports. To facilitate comparison with KWW, we focus on the sample of 57 manufacturing sectors also reported in KWW.

Furthermore, Figure 2 shows the simple relationship between sector level DVS and the processing share, the FIE share in exports, the labor compensation relative to total direct value added, and the capital labor ratio. Figure 2 confirms that sectors that have higher processing share or higher FIEs share in exports tend to have lower DVS. The correlation between DVS and wage share of value added is flat, whereas the correlation between DVS and the capital intensity of the sector is weakly positive. Thus, the more capital-intensive industries are likely to have higher DVS, a result that is probably due to industry upgrading.





Note: this figure shows the correlation between sector level DVS (horizontal axis) and the processing share in exports, the FIE share in exports, the labor compensation relative to total direct value added (wage/value-added), and the capital labor ratio (K/L). *Source*: share of processing exports in total Chinese exports, share of FIEs in total exports are from trade statistics, K/L ratio from the ASIP data, wage/value-added from 2007 benchmark I/O table.

Table 7 shows that within processing exports, the domestic content of FIEs often diverge from that of COEs at the sector level. For example, the DVS of processing COEs is substantially larger than that of processing FIEs in sectors such as electronic components, cultural and office equipment, measuring instruments, and metal products, among others. Conversely, the reverse relationship holds for the shipbuilding, chemical fiber, and steel rolling sectors. Meanwhile, within normal export, COEs have higher DVS in the measuring instrument, other communication equipment, and the railroad transport sectors.

Ta	able 7	7: Decon	position	of Domes	tic Content	bv	Sectors	(%)	, 2007
						· · · ·		· · · /	,

Industry CP P $Total$ P $Total$ P $Total$ P $Total$ P P $Total$ P $Total$ P P $Total$ P		MWZ							processing	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	Industry	СР	FP	CN	FN	Total	Р	N	Total	share
Household Audiovisual Apparatus 27.5 23.4 77.3 74.7 30.3 29.6 75.9 32.6 91.0 Cultural and office quipment 37.7 27.6 75.2 75.5 33.1 33.1 33.1 33.1 33.1 33.1 33.1 33.1 33.1 35.3 75.2 45.8 30.6 66.6 73.9 Electronic computer 19.7 42.4 66.2 70.9 42.5 33.0 75.7 33.9 94.7 Measuring Instruments 41.5 34.6 81.0 71.9 42.2 37.8 80.0 45.8 81.3 Petroleam feline and Nuclear Fuel 15.2 37.5 68.1 64.4 35.6 82.0 51.8 64.1 55.0 75.7 52.5 83.8 67.2 13.1 27.6 79.3 75.7 53.8 27.0 81.8 83.4 65.2 Plasitic 31.1 27.6 73.3 74.5 53.8 27.0 83.4 65.2 59.2 <th></th> <th>[1]</th> <th>[2]</th> <th>[3]</th> <th>[4]</th> <th>[5]</th> <th>[6]</th> <th>[7]</th> <th>[8]</th> <th>[9]</th>		[1]	[2]	[3]	[4]	[5]	[6]	[7]	[8]	[9]
Electronic Component 36.8 21.9 7.2.8 7.6.7 31.4 23.1 7.4.1 36.3 90.5 Clutural and Origine equipment 33.3 33.3 37.7 7.6 7.5.2 7.5.3 33.1 7.4.1 80.5 90.5 Generators 31.3 27.0 80.5 71.1 40.4 86.6 73.9 Electronic computer 19.7 42.4 66.2 70.9 42.5 33.0 78.4 80.0 45.8 81.3 Petroleum feline and Nuclear Fuel 15.2 37.5 68.1 69.2 41.5 34.6 81.0 74.4 80.0 45.8 81.8 64.1 Synthetic Materials 40.3 34.2 73.3 64.4 45.6 83.0 71.0 71.7 53.8 80.4 45.2 81.1 81.6 71.7 53.8 81.4 71.7 53.4 80.5 81.4 71.7 53.3 80.4 45.2 53.9 71.8 71.8 71.7 53.4	Household Audiovisual Apparatus	27.5	25.4	77.3	74.7	30.3	29.6	75.9	32.6	91.0
Cultural and office equipment 37.7 27.6 75.2 75.2 75.4 33.4 33.1 74.1 36.5 90.5 Generators 33.3 27.0 80.5 71.1 40.4 51.2 33.0 76.7 33.9 94.7 Messuring Instruments 19.7 42.4 66.2 70.9 42.5 33.0 75.7 52.8 20.1 68.7 34.4 80.0 45.8 81.3 Petroleum feine and Nuclear Fuel 15.2 35.7 68.1 69.2 45.5 20.1 68.7 74.4 60.4 35.6 81.0 70.9 42.5 31.1 80.8 52.0 51.8 77.7 53.0 80.0 76.7 51.8 81.0 76.7 53.8 82.0 76.7 53.8 83.1 83.1 83.1 83.1 75.7 53.8 83.0 83.4 86.2 55.1 81.7 75.7 53.8 83.0 82.1 85.1 81.7 Readicious canonymanic and opports Acciviviti	Electronic Component	36.8	21.9	72.8	76.7	31.4	23.1	77.5	32.3	83.1
Telecommunication equipment 33.3 33.2 74.7 66.8 39.6 35.3 75.2 43.6 83.1 Electronic computer 19.7 42.4 66.2 70.9 42.5 33.0 75.7 33.9 94.7 Measuring Instruments 41.5 37.4 64.0 71.9 43.2 37.8 80.0 45.8 81.3 Petroleum feline and Nuclear Fuel 15.2 37.5 68.1 69.2 45.5 20.1 81.8 64.1 Syntheric Materials 40.3 34.2 75.3 69.4 47.6 34.0 76.4 47.6 76.4 47.6 83.0 51.8 64.1 Syntheric Materials 71.3 17.3 83.3 52.1 81.1 87.6 77.5 53.8 70.8 81.0 82.1 81.4 51.8 76.5 75.9 53.0 83.1 83.1 81.2 74.4 54.7 74.0 82.0 51.0 74.5 53.0 82.1 41.2 Other social industrial equipment 74.8 83.0 75.0 74.6 75.7 <t< td=""><td>Cultural and office equipment</td><td>37.7</td><td>27.6</td><td>75.2</td><td>75.5</td><td>33.1</td><td>33.1</td><td>74.1</td><td>36.5</td><td>90.5</td></t<>	Cultural and office equipment	37.7	27.6	75.2	75.5	33.1	33.1	74.1	36.5	90.5
Generators 31.3 27.0 80.5 71.1 40.4 51.2 33.0 75.7 33.9 94.7 Messuring Instruments 41.5 34.6 81.0 71.9 42.2 33.8 0.0 45.8 81.3 Petroleum feline and Nuclear Fuel 15.2 37.5 68.1 0.92 45.5 30.0 75.7 52.8 78.2 46.4 35.6 84.0 76.4 43.6 64.1 57.7 53.8 75.7 53.8 75.1 81.9 43.8 86.7 Synhetic Materials 31.1 27.6 79.3 75.7 53.8 71.1 80.8 53.1 81.7 75.7 Rubber 23.2 31.1 81.5 71.7 53.8 27.0 83.4 86.2 51.7 43.0 83.8 51.8 51.7 Ruber Culture, Education and Sports Activities 39.0 80.8 80.0 82.0 64.5 75.6 93.0 80.8 71.4 50.4 50.7 50.7	Telecommunication equipment	33.3	33.2	74.7	66.8	39.6	35.3	75.2	43.6	83.1
Electronic computer 19.7 42.4 66.2 70.9 42.5 33.0 94.7 Measuring Instruments 14.5 37.6 81.4 80.0 45.8 81.3 Petroleum feline and Nucclar Fuel 15.2 37.5 68.1 69.2 45.5 20.1 81.5 Synthetic Materials 26.8 27.2 82.3 67.4 35.6 82.4 47.6 34.0 39.1 83.9 83.0 51.8 64.1 Synthetic Materials 31.1 27.6 73.3 77.7 53.8 27.0 81.8 53.1 81.8 Articles for Culture, Education and Sports Activities 34.3 46.4 32.7 43.0 82.2 44.6 53.0 83.0 51.0 Other special industrial equipment 47.8 50.0 85.0 82.0 64.6 57.6 85.5 92.0 60.8 Other electronic and communication equipment 51.8 51.0 84.5 73.3 65.3 40.0 76.7 33.0 82.0 <td>Generators</td> <td>31.3</td> <td>27.0</td> <td>80.5</td> <td>71.1</td> <td>40.4</td> <td>51.2</td> <td>80.3</td> <td>66.6</td> <td>73.9</td>	Generators	31.3	27.0	80.5	71.1	40.4	51.2	80.3	66.6	73.9
Measuring Instruments 41.5 34.6 81.0 71.9 43.2 37.8 80.0 45.8 81.3 Household electric appliances 26.8 27.2 82.3 78.2 46.4 35.0 68.7 77.7 Ship building 30.7 65.2 84.8 74.7 50.1 80.9 43.8 80.7 75.7 53.8 20.1 80.8 53.1 80.8 53.1 80.8 53.1 80.8 53.1 80.8 53.1 80.8 53.1 80.8 53.1 80.8 53.1 75.7 53.8 75.7 53.8 75.9 <t< td=""><td>Electronic computer</td><td>19.7</td><td>42.4</td><td>66.2</td><td>70.9</td><td>42.5</td><td>33.0</td><td>75.7</td><td>33.9</td><td>94.7</td></t<>	Electronic computer	19.7	42.4	66.2	70.9	42.5	33.0	75.7	33.9	94.7
Petroleum feline and Nuclear Fuel 15.2 37.5 68.1 69.2 45.5 20.1 68.7 44.4 50.6 Synthetic Materials 40.3 34.2 75.3 69.4 47.6 34.0 76.4 47.7 67.7 Ship building 39.7 65.2 84.8 74.7 50.1 39.1 80.3 51.8 51.8 51.8 51.8 51.8 51.8 51.8 51.8 51.8 51.8 51.8 51.8 51.8 51.8 52.5	Measuring Instruments	41.5	34.6	81.0	71.9	43.2	37.8	80.0	45.8	81.3
Household electric appliances 26.8 27.2 82.3 78.2 46.4 35.6 82.0 51.8 64.1 Ship building 39.7 65.2 84.8 74.7 50.1 39.1 83.9 43.8 86.7 Rubber 21.0 77.7 55.3 31.1 80.8 55.1 51.7 Rubber 22.2 31.1 81.5 77.7 53.8 27.0 81.8 53.4 51.8 Other special industrial equipment 46.9 39.1 82.0 74.4 55.7 51.9 76.4 62.6 54.4 Other electroic machinery and equipment 37.8 36.9 79.6 76.9 61.5 33.7 80.3 52.1 41.2 Other electroic and communication equipment 61.8 60.8 81.0 72.7 64.4 34.0 76.7 63.0 62.1 78.1 81.6 71.4 56.9 71.2 30.6 72.7 75.5 69.0 56.1 76.6 71.2 30.6 73.3 66.2 73.3 66.3 76.7 73.3 65.4 <td< td=""><td>Petroleum feline and Nuclear Fuel</td><td>15.2</td><td>37.5</td><td>68.1</td><td>69.2</td><td>45.5</td><td>20.1</td><td>68.7</td><td>44.4</td><td>50.6</td></td<>	Petroleum feline and Nuclear Fuel	15.2	37.5	68.1	69.2	45.5	20.1	68.7	44.4	50.6
Synthetic Materials 40.3 34.2 75.3 69.4 47.6 34.0 76.4 47.7 67.7 Plastic 31.1 27.6 78.3 75.7 52.5 51.1 80.8 55.1 51.7 Plastic 31.1 27.6 79.3 75.7 52.5 81.8 85.34 51.8 Articles for Culture, Education and Sports Activities 34.3 46.4 83.2 74.4 56.4 83.0 51.8 55.2 55.9 Chemical Fiber machinery and equipment 46.8 81.2 72.4 61.4 53.7 80.3 52.1 41.2 Other electric machinery and equipment 61.8 60.8 81.2 72.4 61.4 53.7 80.3 52.1 41.2 Other electric ican communication equipment 61.8 60.8 81.2 72.4 61.4 53.7 80.6 82.0 78.1 70.1 32.7 Special Chemical Products 62.5 51.0 64.5 73.6 66.9 76	Household electric appliances	26.8	27.2	82.3	78.2	46.4	35.6	82.0	51.8	64.1
Ship building39.765.284.874.750.139.183.943.886.7Plastic31.127.679.375.752.531.180.855.151.7Rubber25.231.181.577.553.850.445.645.645.645.645.6Other special industrial equipment34.446.483.774.657.751.975.462.654.4Other electroic and communication equipment71.836.979.676.961.537.780.352.141.2Other electroic and communication equipment61.860.885.082.064.434.780.039.780.6Paper and Paper Products59.050.885.082.064.639.785.170.132.7Special Chemical Products62.551.876.773.368.534.076.761.631.2Nonferrous metal pressing49.357.376.771.456.478.671.230.5Glass and Its Products53.042.287.171.440.490.469.236.7Nonferrous metal smelting49.357.376.771.771.456.486.773.388.6Furniture53.042.287.181.471.664.981.073.873.8Ruber endigment57.057.673.771.456.486.772.336.7 <tr<< td=""><td>Synthetic Materials</td><td>40.3</td><td>34.2</td><td>75.3</td><td>69.4</td><td>47.6</td><td>34.0</td><td>76.4</td><td>47.7</td><td>67.7</td></tr<<>	Synthetic Materials	40.3	34.2	75.3	69.4	47.6	34.0	76.4	47.7	67.7
Plastic 31.1 27.6 79.3 75.7 52.5 31.1 80.8 85.1 51.7 Rubber 25.2 31.1 81.5 77.5 53.8 27.0 81.8 53.4 66.2 Other special industrial equipment 46.9 39.1 82.9 74.6 57.7 43.0 82.5 65.2 55.9 Chemical Fiber 38.0 51.0 76.2 74.8 59.7 43.0 82.5 65.2 55.9 Other electric machinery and equipment 61.8 60.8 81.2 77.4 64.6 57.6 85.5 60.2 60.8 Metal products 39.1 30.5 85.0 82.0 64.6 57.6 85.5 60.2 30.7 88.6 60.8 Metal products 35.3 11.1 83.0 71.1 40.4 90.2 36.7 30.5 81.0 76.7 75.5 69.0 56.1 76.2 34.2 Storerous metal pressing 41.1 53.0 71.4 86.4 76.2 73.3 18.6 77.7 75.6 69.	Ship building	39.7	65.2	84.8	74.7	50.1	39.1	83.9	43.8	86.7
Rubber25.231.181.577.553.853.453.451.8Articles for Culture, Education and Sports Activities3446.483.274.445.643.085.466.2Other special industrial equipment38.051.076.274.857.751.976.462.654.4Other electric machinery and equipment61.861.881.277.464.434.780.352.141.2Other electric machinery and equipment61.860.881.277.464.434.780.639.780.6Paper and Paper Products59.050.885.082.366.939.785.170.132.7Special Chemical Products62.551.876.773.868.534.076.761.631.230.5Special Chemical Products63.551.083.077.777.569.068.534.076.273.874.230.5Glass and Its Products63.042.287.184.771.666.186.776.234.2Other transport equipment57.057.073.271.456.476.273.386.3Paritices59.860.273.873.574.654.981.073.873.6Other transport equipment58.762.181.167.371.974.176.037.0Other manufacturing products59.860.273.673.5 <t< td=""><td>Plastic</td><td>31.1</td><td>27.6</td><td>79.3</td><td>75.7</td><td>52.5</td><td>31.1</td><td>80.8</td><td>55.1</td><td>51.7</td></t<>	Plastic	31.1	27.6	79.3	75.7	52.5	31.1	80.8	55.1	51.7
Articles for Culture, Education and Sports Activities 34.3 46.4 83.2 74.4 56.4 83.0 88.4 66.2 Other special industrial equipment 38.0 51.0 76.2 74.8 59.7 45.0 82.5 65.2 55.9 Other electronic and communication equipment 61.8 60.8 81.2 72.4 64.4 34.7 68.0 39.7 80.6 52.1 41.2 Other electronic and communication equipment 61.8 60.8 81.2 72.4 64.4 34.7 68.0 39.7 80.6 Paper and Paper Products 59.0 50.8 85.0 82.0 64.6 57.6 85.5 69.2 60.8 Metal products 62.5 51.8 76.5 73.3 68.5 34.0 76.7 61.6 31.2 Nonferrous metal prosting 41.1 53.0 71.8 47.1 84.4 70.6 76.7 73.3 64.0 76.2 73.3 18.4 71.6 56.1 86.7 72.3 38.4 Deter instructure quarityment 57.3 76.7 75.7 <td>Rubber</td> <td>25.2</td> <td>31.1</td> <td>81.5</td> <td>77.5</td> <td>53.8</td> <td>27.0</td> <td>81.8</td> <td>53.4</td> <td>51.8</td>	Rubber	25.2	31.1	81.5	77.5	53.8	27.0	81.8	53.4	51.8
Other special industrial equipment46.939.182.974.657.751.976.462.655.4Other electric machinery and equipment37.836.979.676.961.533.780.352.141.2Other electric machinery and equipment61.860.881.272.464.434.768.039.780.6Paper and Paper Products39.130.585.082.366.939.785.170.132.7Special Chemical Products62.551.876.573.368.534.076.761.631.2Nonferrous metal pressing41.153.077.777.569.056.178.671.230.5Glass and Its Products63.551.084.578.078.871.464.494.469.236.7Nonferrous metal smelting49.357.376.763.771.456.476.233.884.5Furniture53.042.287.184.771.656.186.776.234.2Other transport equipment57.050.282.373.673.972.95.4Railroad transport equipment58.762.181.167.371.954.177.769.037.0Other transport equipment59.860.273.675.973.253.673.972.95.4Railroad transport equipment59.860.273.673.973.973.9	Articles for Culture, Education and Sports Activities	34.3	46.4	83.2	74.4	56.4	45.6	83.0	58.4	66.2
$ \begin{array}{l c c c c c c c c c c c c c c c c c c c$	Other special industrial equipment	46.9	39.1	82.9	74.6	57.7	43.0	82.5	65.2	55.9
Other electric machinery and equipment 57.8 36.9 79.6 76.9 61.2 53.7 80.3 52.1 41.2 Paper and Paper Products 59.0 50.8 85.0 82.0 64.4 37.7 88.5 69.2 60.8 Metal products 59.0 50.8 85.0 82.3 66.9 39.7 85.1 70.1 32.7 Special Chemical Products 62.5 51.8 76.5 73.3 68.5 34.0 76.7 61.6 31.2 Sonferrous metal pressing 41.1 53.0 71.7 77.5 69.0 85.1 71.4 64.4 76.2 73.3 18.6 Furniture 53.0 42.2 87.1 84.7 71.4 56.4 76.2 34.2 Other transport equipment 57.0 50.2 82.3 73.6 71.6 54.9 81.0 73.3 28.0 Railroad transport equipment 57.0 50.2 82.1 71.4 54.4 72.6 48.1	Chemical Fiber	38.0	51.0	76.2	74.8	59.7	51.9	76.4	62.6	54.4
Other electronic and communication equipment 61.8 60.8 81.2 72.4 64.4 34.7 68.0 39.7 80.6 Metal products 39.1 30.5 85.0 82.3 66.9 39.7 85.1 70.1 32.7 Special Chemical Products 62.5 51.8 76.5 73.3 68.5 34.0 76.7 61.6 31.2 Nonferrous metal pressing 41.1 53.0 77.7 77.5 69.0 56.1 78.6 71.7 38.4 Leather, fur, down and related products 53.3 41.1 89.3 89.1 71.1 40.4 90.4 69.2 36.7 Nonferrous metal smelting 49.3 57.3 76.7 63.7 71.6 56.1 86.7 76.2 34.2 Other transport equipment 57.0 50.2 82.3 73.8 71.6 54.9 81.0 73.8 28.0 Railroad transport equipment 57.0 50.2 85.6 84.4 72.6 53.6 73.	Other electric machinery and equipment	37.8	36.9	79.6	76.9	61.5	33.7	80.3	52.1	41.2
Paper and Paper Products59.050.880.0 <td>Other electronic and communication equipment</td> <td>61.8</td> <td>60.8</td> <td>81.2</td> <td>72.4</td> <td>64.4</td> <td>34.7</td> <td>68.0</td> <td>39.7</td> <td>80.6</td>	Other electronic and communication equipment	61.8	60.8	81.2	72.4	64.4	34.7	68.0	39.7	80.6
Metai products 39.1 30.5 85.0 82.3 66.9 39.7 85.1 70.1 32.7 Special Chemical Products 62.5 51.8 76.5 73.3 68.5 34.0 76.7 61.6 31.2 Glass and Its Products 35.3 41.1 89.3 71.1 40.4 90.4 69.2 36.7 78.4 Nonferrous metal smelting 49.3 57.3 76.7 63.7 71.4 56.4 76.2 73.3 18.6 Furniture 53.0 42.2 87.1 84.7 71.6 54.1 77.7 75.0 60.2 73.3 18.6 77.0 70.0 70.2 71.4 56.1 86.7 76.2 34.2 Other transport equipment 57.0 50.2 82.3 73.8 71.6 54.9 81.0 73.8 28.0 73.0 70.0 70.0 70.2 53.6 73.3 74.8 84.0 75.3 36.3 72.9 5.4 74.0 84.0 75.3 36.3 72.9 5.4 74.0 84.0 75.3 56.2 </td <td>Paper and Paper Products</td> <td>59.0</td> <td>50.8</td> <td>85.0</td> <td>82.0</td> <td>64.6</td> <td>57.6</td> <td>85.5</td> <td>69.2</td> <td>60.8</td>	Paper and Paper Products	59.0	50.8	85.0	82.0	64.6	57.6	85.5	69.2	60.8
Special Chemical Products 62.5 51.8 76.5 73.5 69.0 76.7 61.6 31.2 Glass and Its Products 63.5 51.0 84.5 78.0 70.8 59.0 83.3 76.7 38.4 Leather, fur, down and related products 33.3 41.1 85.3 89.1 71.1 40.4 90.4 69.2 36.7 Nonferrous metal smelting 49.3 57.3 76.7 63.7 71.4 56.4 76.2 73.4 Cher transport equipment 57.0 70.2 82.3 73.8 71.6 54.1 87.7 76.7 36.7 Other transport equipment 58.7 62.1 81.1 67.3 71.9 54.1 77.7 69.0 37.0 Other manufacturing products 59.8 60.2 73.6 73.5 74.4 84.0 75.3 23.5 36.3 Basic Chemical Raw Materials 28.6 54.3 81.5 72.4 73.5 54.2 83.6 75.6 28.6 75.6 28.6 75.6 28.6 75.6 28.6 75.6 <t< td=""><td>Metal products</td><td>39.1</td><td>30.5</td><td>85.0</td><td>82.3</td><td>66.9</td><td>39.7</td><td>85.1</td><td>70.1</td><td>32.7</td></t<>	Metal products	39.1	30.5	85.0	82.3	66.9	39.7	85.1	70.1	32.7
Nonterrous metal pressing 41.1 5.30 71.7 71.5 69.0 50.1 78.6 71.2 30.5 Class and Its Products 53.3 67.1 88.4 71.1 40.4 90.4 69.2 36.7 Nonferrous metal smelting 49.3 57.3 76.7 63.7 71.4 56.4 76.2 73.3 18.6 Furniture 53.0 42.2 87.1 84.7 71.6 56.4 76.2 34.2 Other transport equipment 57.0 50.2 82.3 73.8 71.6 54.9 81.0 73.8 28.0 Railroad transport equipment 58.7 60.2 73.6 73.5 74.4 86.5 72.3 36.3 Other manufacturing products 59.8 60.2 73.6 73.5 47.4 84.0 73.5 47.4 84.0 73.5 47.4 84.0 73.5 23.6 73.9 75.6 23.8 74.9 16.0 Other industrial machinery 50.3	Special Chemical Products	62.5	51.8	76.5	73.3	68.5	34.0	76.7	61.6	31.2
Glass and its Products 63.5 51.0 84.5 78.0 70.8 85.0 86.7 78.4 Leather, fur, down and related products 35.3 41.1 89.3 89.1 71.1 40.4 90.4 69.2 36.7 Nonferrous metal smelting 49.3 57.3 76.7 63.7 71.4 56.4 76.2 73.3 18.6 Furniture 57.0 50.2 82.3 73.8 71.6 54.9 81.0 73.8 28.0 Railroad transport equipment 58.7 62.1 81.1 67.3 71.9 54.1 77.3 28.0 Other manufacturing products 39.8 53.5 86.6 84.4 72.6 48.1 86.5 72.3 36.3 Pesticides 59.8 60.2 73.6 73.5 47.4 84.0 75.3 23.5 Basic Chemical Raw Materials 28.6 54.3 81.5 72.4 73.5 42.5 80.8 74.9 16.0 Other industrial machinery 50.3 48.7 84.7 80.0 73.5 56.2 83.6	Nonferrous metal pressing	41.1	53.0	77.7	77.5	69.0	56.1	78.6	71.2	30.5
Leather, fur, down and related products 35.3 41.1 89.3 89.1 71.1 40.4 90.4 69.2 36.7 Nonferrous metal smelting 49.3 57.3 76.7 63.7 71.4 56.4 76.2 73.3 18.6 Furniture 57.0 50.2 82.3 73.8 71.6 56.1 86.7 70.2 34.2 Other transport equipment 57.0 50.2 82.3 73.8 71.6 56.1 86.7 73.9 72.9 54.4 Other manufacturing products 39.8 53.5 86.6 84.4 72.6 48.1 86.5 72.3 36.3 Pesticides 59.8 60.2 73.6 73.5 47.4 84.0 75.3 23.5 Motor vehicles 43.0 45.3 83.4 79.6 73.5 47.4 84.0 75.3 23.5 56.2 83.6 75.4 83.6 75.6 28.6 89.7 89.5 70.0 29.7 89.6 75.6 28.6 74.4 84.0 75.3 75.6 62.8 75.6 <td< td=""><td>Glass and Its Products</td><td>63.5</td><td>51.0</td><td>84.5</td><td>78.0</td><td>70.8</td><td>59.0</td><td>83.3</td><td>76.7</td><td>38.4</td></td<>	Glass and Its Products	63.5	51.0	84.5	78.0	70.8	59.0	83.3	76.7	38.4
Nonterrous metal smelting49.357.376.767.771.456.476.273.518.5Furniture53.042.287.184.771.656.186.776.234.2Other transport equipment58.762.181.167.371.954.177.769.037.0Other manufacturing products39.855.586.684.472.648.186.572.336.3Pesticides59.860.273.675.973.253.673.972.95.4Motor vehicles43.045.383.479.673.547.484.075.323.5Basic Chemical Raw Materials28.654.381.572.473.542.580.874.916.0Other industrial machinery50.348.784.780.073.556.283.675.628.6Wearing aparel36.933.952.880.470.275.556.075.61.1Iron-smelting40.545.576.873.975.750.675.61.1Agriculture, forestry, and fishing machinery63.962.382.673.076.257.780.675.622.1Chemical Products for Daily Use64.866.879.981.277.553.375.775.60.4Woolen textiles61.964.090.477.575.381.077.989.889.976.939.8<	Leather, fur, down and related products	35.3	41.1	89.3	89.1	71.1	40.4	90.4	69.2	36.7
Furnitive53.042.287.184.171.656.186.776.234.2Other transport equipment57.050.282.373.871.654.981.073.828.0Railroad transport equipment58.762.181.167.371.954.177.769.037.0Other manufacturing products59.860.273.675.973.253.673.972.95.4Motor vehicles43.045.383.479.673.547.484.075.323.5Basic Chemical Raw Materials28.654.381.572.473.542.580.874.916.0Other industrial machinery50.348.784.780.073.552.883.675.628.6Wearing apparel36.939.389.788.274.353.989.579.029.7Rolling of Steel33.952.876.873.975.750.675.61.1Boiler, engines and turbine59.163.782.371.475.338.781.670.622.1Iron-smelting40.545.576.875.750.675.62.11.1Agriculture, forestry, and fishing machinery63.962.382.676.745.888.078.921.6Smelting of Ferroally62.457.178.271.757.989.476.939.8Chemical Products for Daily Use<	Nonferrous metal smelting	49.3	57.3	76.7	63.7	71.4	56.4	76.2	73.3	18.6
Other transport equipment 57.0 50.2 82.3 71.6 54.9 81.0 73.8 28.0 Railroad transport equipment 58.7 62.1 81.1 67.3 71.9 54.1 77.7 69.0 37.0 Other manufacturing products 39.8 53.5 86.6 84.4 72.6 48.1 86.5 72.3 36.3 Pesticides 43.0 45.3 83.4 79.6 73.5 47.4 84.0 75.3 23.5 Basic Chemical Raw Materials 28.6 54.3 81.5 72.4 73.5 56.2 80.8 74.9 16.0 Other industrial machinery 50.3 48.7 84.7 84.7 73.5 74.3 53.9 89.5 79.0 29.7 Rolling of Steel 33.9 52.8 80.4 70.2 75.3 78.4 70.6 25.0 17.8 83.3 Boiler, engines and turbine 59.1 63.7 82.3 71.4 75.3 38.7 81.6 70.6 25.0 17.0 75.6 25.1 11.1 Agricul	Furniture	53.0	42.2	8/.1	84.7	71.6	56.1	86.7	76.2	34.2
Rairoad transport equipment58.7 62.1 81.1 67.3 71.9 54.1 77.7 69.0 37.0 Other manufacturing products 39.8 53.5 86.6 84.4 72.6 81.1 88.5 72.3 36.3 Pesticides 59.8 60.2 73.6 75.9 73.2 53.6 73.9 72.9 5.4 Motor vehicles 43.0 45.3 83.4 79.6 73.5 47.4 84.0 75.3 22.5 Basic Chemical Raw Materials 28.6 54.3 81.5 72.4 73.5 47.4 84.0 75.3 22.5 Other industrial machinery 50.3 48.7 84.7 80.0 73.5 56.2 83.6 75.6 28.6 Wearing apparel 36.9 39.3 89.7 82.7 73.5 52.9 80.0 77.8 83.3 Boiler, engines and turbine 59.1 63.7 82.3 71.4 75.3 38.7 81.6 70.6 25.0 Iron-smelting 40.5 45.5 76.8 73.0 75.7 50.6 75.6 1.1 Agriculture, forestry, and fishing machinery 63.4 66.8 79.9 81.5 76.3 58.4 80.8 73.3 30.0 Cotin textiles 61.9 64.0 90.4 79.3 77.7 57.9 89.4 76.9 39.8 Fertilizers 55.7 60.0 82.8 78.2 78.4 76.8 81.0 <td>Other transport equipment</td> <td>57.0</td> <td>50.2</td> <td>82.3</td> <td>73.8</td> <td>71.6</td> <td>54.9</td> <td>81.0</td> <td>73.8</td> <td>28.0</td>	Other transport equipment	57.0	50.2	82.3	73.8	71.6	54.9	81.0	73.8	28.0
Other manufacturing products 39.8 53.5 80.6 84.4 72.6 48.1 86.5 72.3 36.3 Pesticides 59.8 60.2 73.6 73.5 73.7 73.6 73.9 72.9 5.4 Motor vehicles 43.0 45.3 83.4 79.6 73.5 47.4 84.0 75.3 23.5 Basic Chemical Raw Materials 28.6 54.3 81.5 72.4 73.5 42.5 80.8 74.9 16.0 Other industrial machinery 50.3 48.7 84.7 70.0 73.5 56.2 83.6 75.6 28.6 Wearing apparel 36.9 39.3 89.7 88.2 74.3 53.9 89.5 79.0 29.7 Rolling of Steel 33.9 52.8 80.4 70.2 75.2 52.9 80.0 77.8 8.3 Boiler, engines and turbine 63.7 62.3 82.6 73.0 76.2 57.7 80.6 75.6 1.1 Agriculture, forestry, and fishing machinery 63.9 62.3 82.6 73.0	Railroad transport equipment	58.7	62.1	81.1	67.3	71.9	54.1	77.7	69.0	37.0
Pesticides59.860.273.973.253.673.972.95.4Motor vehicles43.045.383.479.673.547.484.075.323.5Basic Chemical Raw Materials28.654.381.572.473.542.580.874.916.0Other industrial machinery50.348.784.780.073.556.283.675.628.6Rolling of Steel36.939.389.788.274.353.989.579.029.7Rolling of Steel33.952.880.470.275.252.980.077.88.3Boiler, engines and urbine59.163.782.371.475.338.781.670.625.0Iron-smelting40.545.576.873.076.257.780.675.622.1Chemical Products for Daily Use64.866.879.981.576.358.480.873.330.0Cotton textiles30.043.387.986.976.745.888.078.921.6Smelting of Ferroalloy62.457.178.278.478.357.756.60.4Woole textiles61.964.090.479.377.757.989.476.939.8Fertilizers65.760.082.878.278.456.875.672.639.0Printing Inks, Pigments and Similar Products63.0	Other manufacturing products	39.8	53.5	86.6	84.4	72.6	48.1	86.5	72.3	36.3
Motor vehicles43.043.383.473.573.541.484.075.325.5Basic Chemical Raw Materials28.654.381.572.473.542.580.874.916.0Other industrial machinery50.348.784.780.073.556.283.675.628.6Wearing apparel36.939.389.788.274.353.989.579.029.7Rolling of Steel33.952.880.470.275.252.980.077.88.3Boiler, engines and turbine59.163.782.371.475.338.781.670.625.0Iron-smelting40.545.576.873.975.750.675.61.1Agriculture, forestry, and fishing machinery63.962.382.673.076.257.780.675.622.1Chemical Products for Daily Use64.866.879.981.576.358.480.873.330.0Cotto textiles30.043.387.986.976.745.888.078.921.6Smelting of Ferroalloy62.457.178.271.575.377.757.60.4Woolen textiles61.964.090.479.377.757.989.476.939.8Fertilizers57.660.082.872.478.357.381.077.316.0Piniting, Reproduction of Re	Pesticides	59.8	60.2	73.6	75.9	73.2	53.6	73.9	72.9	5.4
Basic Chemical Raw Materials28.654.381.572.473.542.580.874.916.0Other industrial machinery50.348.784.780.073.556.283.675.628.6Wearing apparel36.939.389.788.274.353.989.579.029.7Rolling of Steel33.952.880.470.275.252.980.077.88.3Boiler, engines and turbine59.163.782.371.475.338.781.670.625.0Iron-smelting40.545.576.873.975.750.675.61.1Agriculture, forestry, and fishing machinery63.962.382.673.076.257.780.675.622.1Chemical Products for Daily Use64.866.879.981.577.553.375.775.60.4Smelting of Feroalloy62.457.178.271.577.337.775.60.4Woolen textiles61.964.090.479.377.757.989.476.939.8Fertilizers55.760.082.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.461.682.878.278.456.881.277.316.0Printing, Reproduction of Recording Media63.068.886.985.579.061.086.4	Motor vehicles	43.0	45.3	83.4	79.6	73.5	47.4	84.0	75.3	23.5
Other industrial machinery 50.3 48.7 80.0 73.5 50.2 85.6 75.6 28.6 Wearing apparel 36.9 39.3 89.7 88.2 74.3 53.9 89.5 79.0 29.7 Rolling of Steel 33.9 52.8 80.4 70.2 75.2 52.9 80.0 77.8 8.3 Boiler, engines and turbine 59.1 63.7 82.3 71.4 75.3 38.7 81.6 70.6 25.0 Iron-smelting 40.5 45.5 76.8 73.0 76.2 57.7 80.6 75.6 22.1 Chemical Products for Daily Use 64.8 66.8 79.9 81.5 76.3 58.4 80.8 73.3 30.0 Cotton textiles 30.0 43.3 87.9 86.9 76.7 45.8 88.0 78.9 21.6 Smelting of Ferroalloy 62.4 57.1 78.2 77.5 53.3 75.7 75.6 0.4 Woolen textiles 61.9 64.0 90.4 79.3 77.7 57.9 89.4 7	Basic Chemical Raw Materials	28.6	54.3	81.5	72.4	73.5	42.5	80.8	74.9	16.0
Wearing apparel36.939.339.389.788.274.353.989.519.029.7Rolling of Steel33.952.880.470.275.252.980.077.88.3Boiler, engines and turbine59.163.782.371.475.338.781.670.625.0Iron-smelting40.545.576.873.975.750.675.975.61.1Agriculture, forestry, and fishing machinery63.962.382.673.076.257.780.675.622.1Chemical Products for Daily Use64.866.879.981.576.358.480.873.330.0Cotton textiles30.043.387.986.976.745.888.078.921.6Smelting of Ferroalloy62.457.178.271.577.553.375.775.60.4Woolen textiles61.964.090.479.377.757.989.476.939.8Fertilizers55.760.082.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.068.886.985.579.061.086.476.539.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.887.978.776.6 <td< td=""><td>Other industrial machinery</td><td>50.3</td><td>48.7</td><td>84.7</td><td>80.0</td><td>73.5</td><td>56.2</td><td>83.6</td><td>75.6</td><td>28.6</td></td<>	Other industrial machinery	50.3	48.7	84.7	80.0	73.5	56.2	83.6	75.6	28.6
Koling of Steel53.952.880.470.275.252.980.077.88.3Boiler, engines and turbine59.163.782.371.475.338.781.670.625.0Iron-smelting40.545.576.873.975.750.675.975.61.1Agriculture, forestry, and fishing machinery63.962.382.673.076.257.780.675.622.1Chemical Products for Daily Use64.866.879.981.576.358.480.873.330.0Cotto textiles30.043.387.986.976.745.888.078.921.6Smelting of Ferroalloy62.457.178.271.577.553.375.775.60.4Woolen textiles61.964.090.479.377.757.989.476.939.8Fertilizers55.760.082.872.478.357.381.077.913.0Metalworking machinery63.461.682.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products57.461.384.882.180.258.480.416.1Pottery and Porcelain65.063.886.985.579.061.086.476.539.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.251.680.4 <td>Wearing apparel</td> <td>36.9</td> <td>39.3</td> <td>89.7</td> <td>88.2</td> <td>74.3</td> <td>53.9</td> <td>89.5</td> <td>79.0</td> <td>29.7</td>	Wearing apparel	36.9	39.3	89.7	88.2	74.3	53.9	89.5	79.0	29.7
Boiler, engines and turbine59.163.782.371.475.358.781.670.625.0Iron-smelting40.545.576.873.975.750.675.975.61.1Agriculture, forestry, and fishing machinery63.962.382.673.076.257.780.675.622.1Chemical Products for Daily Use64.866.879.981.576.358.480.873.330.0Cotton textiles30.043.387.986.976.745.888.078.921.6Smelting of Ferroalloy62.457.178.271.577.553.375.775.60.4Woolen textiles61.964.090.479.377.757.989.476.939.8Fertilizers55.760.082.872.478.357.381.077.913.0Metalworking machinery63.461.682.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.068.886.985.579.061.086.476.539.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.7 <t< td=""><td>Rolling of Steel</td><td>33.9</td><td>52.8</td><td>80.4</td><td>70.2</td><td>75.2</td><td>52.9</td><td>80.0</td><td>//.8</td><td>8.3</td></t<>	Rolling of Steel	33.9	52.8	80.4	70.2	75.2	52.9	80.0	//.8	8.3
Iron-smelting40.545.576.875.975.780.675.975.61.1Agriculture, forestry, and fishing machinery63.962.382.673.076.257.780.675.622.1Chemical Products for Daily Use64.866.879.981.576.358.480.873.330.0Cotton textiles30.043.387.986.976.745.888.078.921.6Smelting of Ferroalloy62.457.178.271.577.553.375.775.60.4Woolen textiles61.964.090.479.377.757.989.476.939.8Fertilizers55.760.082.872.478.357.381.077.913.0Metalworking machinery63.461.682.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.068.886.985.579.061.086.476.539.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.084.4Itertiles productions45.260.188.886.581.254.988.482.076.6Vinter and crocheted fabrics and articles40.849.387.987.0 <t< td=""><td>Boiler, engines and turbine</td><td>59.1</td><td>63.7</td><td>82.3</td><td>71.4</td><td>75.3</td><td>38.7</td><td>81.6</td><td>70.6</td><td>25.0</td></t<>	Boiler, engines and turbine	59.1	63.7	82.3	71.4	75.3	38.7	81.6	70.6	25.0
Agriculture, torestry, and insting machinery63.962.382.673.076.257.780.675.622.1Chemical Products for Daily Use64.866.879.981.576.358.480.873.330.0Cotton textiles30.043.387.986.976.745.888.078.921.6Smelting of Ferroalloy62.457.178.271.577.553.375.775.60.4Woolen textiles61.964.090.479.377.757.989.476.939.8Fertilizers55.760.082.872.478.357.381.077.913.0Metalworking machinery63.461.682.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.068.886.985.579.056.876.572.639.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.780.880.80.2Textiles productions45.260.188.886.581.254.988.482.420.7Knitted and crocheted fabrics and articles40.849.387.987.081	Iron-smelting	40.5	45.5	/6.8	73.9	75.7	50.6	/5.9	/5.6	1.1
Chemical Products for Daily Use64.866.879.981.576.358.480.873.330.0Cotton textiles30.043.387.986.976.745.888.078.921.6Smelting of Ferroalloy62.457.178.271.553.375.775.60.4Woolen textiles61.964.090.479.377.757.989.476.939.8Fertilizers55.760.082.872.478.357.381.077.913.0Metalworking machinery63.461.682.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.068.886.985.579.056.876.539.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.780.80.2Textiles productions45.260.188.886.581.251.688.282.515.6Other non-metallic mineral products65.262.787.572.782.556.686.083.011.3Fire-resistant Materials63.160.188.281.382.937.587.680.313.9 <t< td=""><td>Agriculture, forestry, and fishing machinery</td><td>63.9</td><td>62.3</td><td>82.6</td><td>/3.0</td><td>76.2</td><td>57.7</td><td>80.6</td><td>/5.6</td><td>22.1</td></t<>	Agriculture, forestry, and fishing machinery	63.9	62.3	82.6	/3.0	76.2	57.7	80.6	/5.6	22.1
Cotton textules30.043.387.986.976.745.888.078.921.6Smelting of Feroalloy62.457.178.271.553.375.775.60.4Woolen textiles61.964.090.479.377.757.989.476.939.8Fertilizers55.760.082.872.478.357.381.077.913.0Metalworking machinery63.461.682.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.068.886.985.579.061.086.476.539.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.780.880.80.2Textiles productions45.260.188.886.581.254.988.482.420.7Knitted and crocheted fabrics and articles40.849.387.987.081.551.688.282.515.6Other non-metallic mineral products65.262.787.572.782.555.186.684.79.4Hemp textiles60.058.187.983.782.937.587.680.3<	Chemical Products for Daily Use	64.8	66.8	/9.9	81.5	76.3	58.4	80.8	73.3	30.0
Similar periodity62.457.178.271.577.553.375.775.00.4Woolen textiles61.964.090.479.377.757.989.476.939.8Fertilizers55.760.082.872.478.357.381.077.913.0Metalworking machinery63.461.682.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.068.886.985.579.056.876.572.639.0Printing, Reproduction of Recording Media63.068.886.985.579.061.086.476.539.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.780.880.80.2Textiles productions45.260.188.886.581.251.688.282.515.6Other non-metallic mineral products65.262.787.572.782.556.686.083.011.3Fire-resistant Materials63.160.188.281.382.555.186.684.79.4Medicines60.058.187.983.782.937.587.6<	Cotton textiles	30.0	43.3	87.9	86.9	76.7	45.8	88.0	78.9	21.6
Woolen textnes61.964.090.479.377.757.989.476.959.8Fertilizers55.760.082.872.478.357.381.077.913.0Metalworking machinery63.461.682.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.068.886.985.579.056.876.572.639.0Printing, Reproduction of Recording Media63.068.886.985.579.061.086.476.539.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.780.880.80.2Textiles productions45.260.188.886.581.254.988.482.420.7Knitted and crocheted fabrics and articles40.849.387.987.081.551.688.282.515.6Other non-metallic mineral products65.262.787.572.782.556.686.083.011.3Fire-resistant Materials63.160.188.281.382.555.186.684.79.4Medicines60.058.187.983.782.9 </td <td>Smelling of Ferroalloy</td> <td>02.4</td> <td>57.1</td> <td>/8.2</td> <td>/1.5</td> <td>77.5</td> <td>55.5</td> <td>/5./</td> <td>75.0</td> <td>0.4</td>	Smelling of Ferroalloy	02.4	57.1	/8.2	/1.5	77.5	55.5	/5./	75.0	0.4
Fermizers55.760.082.872.478.357.381.077.315.0Metalworking machinery63.461.682.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.068.886.985.579.056.876.572.639.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.780.880.80.2Textiles productions45.260.188.886.581.254.988.482.420.7Knitted and crocheted fabrics and articles40.849.387.987.081.551.688.282.515.6Other non-metallic mineral products65.262.787.572.782.556.686.083.011.3Fire-resistant Materials63.160.188.281.382.555.186.684.79.4Medicines60.058.187.983.782.937.587.680.313.9Hemp textiles64.162.588.080.784.856.886.683.99.9Cement, Lime and Plaster61.243.089.881.786.252.98	woolen textiles	01.9 55 7	64.0	90.4	79.5	77.7	57.9	89.4	70.9	39.8
Metalworking machinery65.461.682.878.278.456.881.277.316.0Paints, Printing Inks, Pigments and Similar Products63.068.886.985.579.056.876.572.639.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.780.880.80.2Textiles productions45.260.188.886.581.254.988.482.420.7Knitted and crocheted fabrics and articles40.849.387.987.081.551.688.282.515.6Other non-metallic mineral products65.262.787.572.782.556.686.083.011.3Fire-resistant Materials63.160.188.281.382.555.186.684.79.4Medicines64.162.588.080.784.856.886.683.99.9Cement, Lime and Plaster61.243.089.881.786.252.989.088.42.8CokingN/AN/A90.873.988.8N/A89.689.60.0	Fertilizers	55./ (2.4	60.0	82.8	72.4	78.3	57.3	81.0	77.9	13.0
Paints, Printing Inks, Prignents and Similar Products65.068.886.985.579.056.876.572.659.0Printing, Reproduction of Recording Media63.068.886.985.579.061.086.476.539.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.780.880.80.2Textiles productions45.260.188.886.581.254.988.482.420.7Knitted and crocheted fabrics and articles40.849.387.987.081.551.688.282.515.6Other non-metallic mineral products65.262.787.572.782.556.686.083.011.3Fire-resistant Materials63.160.188.281.382.555.186.684.79.4Medicines64.162.588.080.784.856.886.683.99.9Cement, Lime and Plaster61.243.089.881.786.252.989.088.42.8CokingN/AN/A90.873.988.8N/A89.689.60.0	Deinte Drinting machinery	03.4	01.0	82.8	18.2	78.4	50.8	81.2	77.5	10.0
Printing, Reproduction of Recording Media65.068.886.985.579.061.080.476.559.0Wood, Bamboo, Rattan, Palm and Straw Products57.461.384.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.780.880.80.2Textiles productions45.260.188.886.581.254.988.482.420.7Knitted and crocheted fabrics and articles40.849.387.987.081.551.688.282.515.6Other non-metallic mineral products65.262.787.572.782.556.686.083.011.3Fire-resistant Materials63.160.188.281.382.555.186.684.79.4Medicines64.162.588.080.784.856.886.683.99.9Cement, Lime and Plaster61.243.089.881.786.252.989.088.42.8CokingN/AN/A90.873.988.8N/A89.689.60.0	Paints, Printing Inks, Pigments and Similar Products	63.0	68.8	86.9	85.5	79.0	56.8	/6.5	72.6	39.0
Wood, Bamboo, Rattan, Pain and Straw Products57.461.584.882.180.258.484.680.416.1Pottery and Porcelain65.063.884.378.281.158.283.482.08.4SteelmakingN/A17.481.776.081.251.780.880.80.2Textiles productions45.260.188.886.581.254.988.482.420.7Knitted and crocheted fabrics and articles40.849.387.987.081.551.688.282.515.6Other non-metallic mineral products65.262.787.572.782.556.686.083.011.3Fire-resistant Materials63.160.188.281.382.555.186.684.79.4Medicines60.058.187.983.782.937.587.680.313.9Hemp textiles64.162.588.080.784.856.886.683.99.9Cement, Lime and Plaster61.243.089.881.786.252.989.088.42.8CokingN/AN/A90.873.988.8N/A89.689.60.0	Printing, Reproduction of Recording Media	63.0 57.4	08.8	80.9	85.5	79.0	59.4	80.4	/0.5	39.0
Pottery and Porcelain 05.0 05.8 84.5 78.2 81.1 58.2 85.4 82.0 8.4 Steelmaking N/A 17.4 81.7 76.0 81.2 51.7 80.8 80.8 0.2 Textiles productions 45.2 60.1 88.8 86.5 81.2 54.9 88.4 82.4 20.7 Knitted and crocheted fabrics and articles 40.8 49.3 87.9 87.0 81.5 51.6 88.2 82.5 15.6 Other non-metallic mineral products 65.2 62.7 87.5 72.7 82.5 56.6 86.0 83.0 11.3 Fire-resistant Materials 63.1 60.1 88.2 81.3 82.5 55.1 86.6 84.7 9.4 Medicines 60.0 58.1 87.9 83.7 82.9 37.5 87.6 80.3 13.9 Hemp textiles 64.1 62.5 88.0 80.7 84.8 56.8 86.6 83.9 9.9 Cement, Lime and Plaster 61.2 43.0 89.8 81.7 86.2<	Wood, Bamboo, Kattan, Paim and Straw Products	57.4	61.3	84.8	82.1	80.2	58.4	84.6	80.4	16.1
Steemaking N/A 17.4 81.7 76.0 81.2 51.7 80.8 80.5 0.2 Textiles productions 45.2 60.1 88.8 86.5 81.2 54.9 88.4 82.4 20.7 Knitted and crocheted fabrics and articles 40.8 49.3 87.9 87.0 81.5 51.6 88.2 82.5 15.6 Other non-metallic mineral products 65.2 62.7 87.5 72.7 82.5 56.6 86.0 83.0 11.3 Fire-resistant Materials 63.1 60.1 88.2 81.3 82.5 55.1 86.6 84.7 9.4 Medicines 60.0 58.1 87.9 83.7 82.9 37.5 87.6 80.3 13.9 Hemp textiles 64.1 62.5 88.0 80.7 84.8 56.8 86.6 83.9 9.9 Cement, Lime and Plaster 61.2 43.0 89.8 81.7 86.2 52.9 89.0 88.4 2.8 Coking N/A N/A 90.8 73.9 88.8 N/A	Pottery and Porcelain	05.U	03.8	84.3	76.2	81.1 91.2	58.2	83.4	82.0	8.4
Textures productions 43.2 00.1 88.8 80.5 81.2 54.9 88.4 82.4 20.7 Knitted and crocheted fabrics and articles 40.8 49.3 87.9 87.0 81.5 51.6 88.2 82.4 20.7 Other non-metallic mineral products 65.2 62.7 87.5 72.7 82.5 56.6 86.0 83.0 11.3 Fire-resistant Materials 63.1 60.1 88.2 81.3 82.5 55.1 86.6 84.7 9.4 Medicines 60.0 58.1 87.9 83.7 82.9 37.5 87.6 80.3 13.9 Hemp textiles 64.1 62.5 88.0 80.7 84.8 56.8 86.6 83.9 9.9 Cement, Lime and Plaster 61.2 43.0 89.8 81.7 86.2 52.9 89.0 88.4 2.8 Coking N/A N/A 90.8 73.9 88.8 N/A 89.6 89.6 0.0	Steennaking Textiles productions	IN/A	1/.4	81./ 00 0	/0.U 86 5	01.2 91.2	54.0	8U.8	80.8 82 4	0.2
Kintee and crochered fabrics and articles 40.8 49.3 87.9 87.0 81.5 51.6 88.2 82.5 15.6 Other non-metallic mineral products 65.2 62.7 87.5 72.7 82.5 56.6 86.0 83.0 11.3 Fire-resistant Materials 63.1 60.1 88.2 81.3 82.5 55.1 86.6 84.7 9.4 Medicines 60.0 58.1 87.9 83.7 82.9 37.5 87.6 80.3 13.9 Hemp textiles 64.1 62.5 88.0 80.7 84.8 56.8 86.6 83.9 9.9 Cement, Lime and Plaster 61.2 43.0 89.8 81.7 86.2 52.9 89.0 88.4 2.8 Coking N/A N/A 90.8 73.9 88.8 N/A 89.6 89.6 0.0	rextiles productions	45.2	0U.1	88.8	80.3 87.0	81.2 91 5	54.9	88.4	82.4 82.5	20.7
Other non-metanic inneral products 05.2 02.7 87.5 72.7 82.5 50.6 80.0 83.0 11.3 Fire-resistant Materials 63.1 60.1 88.2 81.3 82.5 55.1 86.6 84.7 9.4 Medicines 60.0 58.1 87.9 83.7 82.9 37.5 87.6 80.3 13.9 Hemp textiles 64.1 62.5 88.0 80.7 84.8 56.8 86.6 83.9 9.9 Cement, Lime and Plaster 61.2 43.0 89.8 81.7 86.2 52.9 89.0 88.4 2.8 Coking N/A N/A 90.8 73.9 88.8 N/A 89.6 89.6 0.0	Nintee and crocneted fabrics and articles	40.8	49.5	8/.9 07 5	87.0 72.7	01.5 02.7	51.0	86.2	82.3 82.0	15.0
Price-resistant valuenais 05.1 00.1 88.2 81.3 82.5 55.1 80.6 84.7 9.4 Medicines 60.0 58.1 87.9 83.7 82.9 37.5 87.6 80.3 13.9 Hemp textiles 64.1 62.5 88.0 80.7 84.8 56.8 86.6 83.9 9.9 Cement, Lime and Plaster 61.2 43.0 89.8 81.7 86.2 52.9 89.0 88.4 2.8 Coking N/A N/A 90.8 73.9 88.8 N/A 89.6 89.6 0.0	Cuter non-metallic mineral products	62.1	02.7	8/.5	12.1	82.5 82 5	50.0	80.0	83.U	11.5
Medicines 00.0 58.1 87.9 83.7 82.9 57.5 87.6 80.5 13.9 Hemp textiles 64.1 62.5 88.0 80.7 84.8 56.8 86.6 83.9 9.9 Cement, Lime and Plaster 61.2 43.0 89.8 81.7 86.2 52.9 89.0 88.4 2.8 Coking N/A N/A 90.8 73.9 88.8 N/A 89.6 89.6 0.0	File-resistant Materials	03.1	50.1	88.2 87.0	81.3 82.7	04.5	33.1 27 5	80.0	84./	9.4
nemp textures 04.1 02.5 88.0 80.7 84.3 50.8 80.6 83.9 9.9 Cement, Lime and Plaster 61.2 43.0 89.8 81.7 86.2 52.9 89.0 88.4 2.8 Coking N/A N/A 90.8 73.9 88.8 N/A 89.6 89.6 0.0	Medicines	0U.U	58.1	8/.9	83./ 80.7	84.9	31.3 500	8/.0	80.3	13.9
Cement, Line and Plaster 01.2 43.0 89.8 81.7 86.2 52.9 89.0 88.4 2.8 Coking N/A N/A 90.8 73.9 88.8 N/A 89.6 89.6 0.0	Compart, Lime and Plaster	04.1	02.5	88.0	8U./	04.ð	50.8	80.0	83.9 00 4	9.9
<u>Coking</u> N/A N/A 90.8 / 3.9 88.8 N/A 89.6 89.6 0.0	Colting	01.2 N/A	43.U	89.8	ð1./	80.2	52.9 N/A	89.0	88.4	2.8
		IN/A	IN/A	90.8	/3.9	00.0 70.0	IN/A	89.0	89.0	0.0

Note: This table gives the decomposition of domestic content by manufacturing sectors. Columns [1]-[5] present DVS estimated in this paper, while columns [6]-[8] present DVS estimates by KWW. Column [9] gives the share of processing exports in total exports. To facilitate comparison with KWW, we focus on the sample of 57 manufacturing sectors also reported in KWW.

Considering that our findings are worth comparing with KWW at the sector level, we therefore list the DVS values by processing, normal, and their weighted sum in columns (6) to (9), which are shown in Table 6 of KWW. Our method is based on the optimization strategy proposed in KWW; however, we consider ownership within each trade mode. Our estimation further adds restrictions on the balance conditions for COEs and FIEs, which captures the internal difference in DVS by ownership type within each trade mode. Thus, the correlation between our sector level DVS and KWW's is over 0.9. On the other hand, the correlation between our CP's and FP's DVS and KWW's processing export DVS is only 0.65 and 0.67, respectively; the correlation between our CN's and FN's DVS and KWW's normal export DVS is 0.86 and 0.69, respectively.

IV. C. Distribution of export value to factor owners

In the previous subsections, we estimated the domestic contents share in export, for each type of firms at the aggregate and sector levels. These domestic contents, however, may likewise become foreign income because of foreign ownership. Thus, we further study the income distribution of exports across ownership types. We add a row of foreign factor income at the bottom of the expanded I/O table to capture the foreign factor inputs in each sector (i.e., Table 2). Two types of foreign factor inputs in domestic production exist, namely, capital and labor. As discussed in the data section, we utilize the merged sample to obtain the value added and foreign share in total paid-in capital for the four types of firms to estimate foreign capital income from domestic contents in exports. In addition, we collect sector-level investment income data from the BOP table compiled by the People's Bank of China. In each sector, we apply the labor and capital income share based on the BOP to compute for the share of foreign factor in gross exports by four types of firms.²³

Table 8 presents our estimates on foreign income shares in processing and normal exports by either COEs or FIEs, which are based on Equations (12) to (14). For reference, we first present the total DVS and FVS in exports for each type of firms in the first two rows, followed by the foreign income share in domestic content in row (3). A small share of foreign income was

²³ The sector-level results of foreign income share are listed in the online Appendix C (Table C3).

found in the exports generated by COEs. For each 100 USD of processing exports by COEs in 2007, 35.5 USD are generated by domestic production factors, only 0.5 USD can be attributed to foreign factor income [column (1)]. Similarly, for each 100 USD of normal exports by COEs, 84 USD are generated by domestic factors, only 0.76 USD is attributed to foreign factor income [column (3)], whereas the foreign factor income share in FIEs' exports is much higher. For example, for each 100 USD processing exports by FIEs, 37 dollars are generated by domestic factors, whereas 16.3 USD can be attributed to foreign factor income [column (2)]. Similarly, for each 100 USD of normal exports by FIEs, 79.5 USD are generated by domestic factors, and a striking 32.7 USD is attributed to foreign factor income [column (4)].

To estimate the domestic GNI of exports, we can use domestic content in exports minus the part that goes to foreign factors income. Then, by using foreign content in exports²⁴ plus the foreign income from domestic content in exports, we can obtain the foreign GNI of Chinese exports. The results are shown in the last two rows of Table 8. The majority of processing exports in China contribute to foreign GNI, regardless of the ownership type of exporters. For example, for 100 USD processing exports by COEs, 65 USD go to foreign GNI, and 79 USD go to processing exports by FIEs. About 53.2 percent of normal exports by FIEs can be attributed to foreign GNI, whereas the foreign GNI share in normal exports by COEs is much lower, which is around 16.6 percent. Overall, there are 47.4 USD in domestic GNI and 52.6 USD in foreign GNI for each 100 USD in Chinese gross exports [column (5)].

	Processing exports by		Normal exports By		Aggregate
	COEs	FIEs	COEs	FIE	
Total DVS	35.46	37.30	84.11	79.53	59.17
Total FVS	64.54	62.7	15.89	20.47	40.83
Total Foreign Income Share in Domestic Value-added (TFI)	0.50	16.34	0.76	32.74	11.78
TDNI	34.96	20.96	83.35	46.79	47.39
TFNI	65.04	79.04	16.65	53.21	52.61

Table 8: National Income and Foreign Income Share in China's Exports, 2007 (%)

Note: This table presents estimates on foreign income shares in processing and normal exports by either COEs or FIEs. TFI denotes total foreign incomes in total domestic value added, TDNI and TFNI denotes total domestic national incomes and foreign incomes of gross exports.

²⁴ We assume that no Chinese-owned factor income are found in these imported foreign value-added, although we admit that China's outgoing FDI has grown in recent years.

The distribution of gross export income is summarized in Figure 3. For each of the 100 USD gross exports, on average 59.17 USD goes to domestic source and 40.83 USD goes to foreign source. Domestic sourced share is higher. For the 59.17 USD of domestic content in gross exports, 32.35 USD is attributed to the COEs, and 26.82 USD is attributed to the FIEs. The foreign factor owners in COEs receive a negligible income (0.31 USD) out of the 32.35 USD of domestic sourced value by COEs, and foreign factor owners in FIEs receive 11.47 USD worth of income out of the 26.82 USD of domestic-sourced value by FIEs. Adding the 40.83 USD of foreign-sourced value, the total foreign income is 52.61 USD, and the total domestic income is 47.39 USD. Thus, in terms of income, domestic share is lower than domestic content share as well.





Note: this figure describes how 100 USD gross exports from China can be decomposed into domestic and foreign content, by Chinese-owned enterprises (COEs) and foreign-owned enterprises (FIEs), and finally the distribution of gross export income into factors of different nationality.

Source: Based on authors' calculation.

IV.D. Robustness check

In our benchmark estimation, we set the initial values for inter-industry, inter-firm type transaction, final demand, and value added, $z0_j^{lk}$, $y0_j^{lk}$, and $v0_j^{lk}$, based on the available official I/O table and other available statistics. However, different sets of initial values may affect our estimated I/O table by firm types. In addition to the above-mentioned initial values, we performed three alternatives in order to test the sensitivity of our results.

For the first alternative, we adjust the initial value for the direct value added, $v0_j^k$, as the residual of the total gross output minus total intermediate input:

$$v0_{j}^{lk} = x0_{j}^{lk} - \sum_{o} \sum_{i=1}^{K} z0_{ij}^{olk} - \sum_{i=1}^{K} z0_{ij}^{mlk} \quad \text{where } o, l = C, F \quad \&k = N, P$$
(28)

Second, we adjust the usage of intermediate inputs by imposing proportionality assumption. Thus, the intermediate input matrix is set as follows:

$$z O_{ij}^{olk} = \left[(x_i^o - e_i^o - y O_i^o) / (x_i - e_i - y_i) \right] \left[(x O_j^{lk} - v O_j^{lk}) / (x_j - v_j) \right] z_{ij}$$
(29)

$$z O_{ij}^{mlk} = \left[(m_i - y_i^m) / (x_i - e_i - y_i) \right] \left[(x O_j^{lk} - v O_j^{lk}) / (x_j - v_j) \right] z_{ij}$$
(30)

where $[(x0_j^{lk} - v0_j^{lk})/(x_j - v_j)]$ is the proportion of intermediate inputs used in sector *j* by firms of type *lk*, relative to total input usage in sector *j*. $[(x_i^o - e_i^o - y0_i^o)/(x_i - e_i - y_i)]$ is the proportion of input production by firms with ownership *o* (i.e., FIEs or COEs) in sector *i* relative to the total input production in sector *i*. Likewise, $[(m_i - y_i^m)/(x_i - e_i - y_i)]$ is the imported inputs in sector *i* relative to the total input production in sector *i*.

In the final robustness check, we simultaneously adjust the initial value of value added and input matrix, according to Equations (28) to (30). Table 9 lists the results of direct domestic value added and total domestic content share using different initial values. All alternative initializations provide qualitatively and quantitatively similar patterns and magnitudes. Thus, we conclude that our benchmark estimation is robust, as long as we control for the gross output and value added for each industry by COEs and FIEs from industrial survey data, as well as intermediate imports used for each type of firm from trade statistics.²⁵

²⁵ Sector level results of these sensitivity tests are listed in the online Appendix C (Table C4). Except for a few cases, most sectors are still in the same range of total DVS as the benchmark. The correlation between benchmark and alternative method 1, for example, is 95%.

		Process	Processing exports		Normal exports	
			by		By	
	Scenario	COEs	FIEs	COEs	FIEs	
Direct domestic content	benchmark	15.58	16.64	22.14	23.00	19.33
	alternative 1	11.43	16.32	22.30	23.22	18.94
	alternative 2	8.25	10.70	22.40	25.20	16.53
	alternative 3	8.58	15.77	22.46	23.77	18.60
Total domestic content	benchmark	35.46	37.30	84.11	79.53	59.17
	alternative 1	34.49	37.26	84.27	79.00	59.06
	alternative 2	34.49	36.72	84.25	79.93	58.94
	alternative 3	33.89	37.14	84.34	79.11	59.00

Table 9: Robustness Check on Direct DVS and Total DVS Using Different Initial Values.

Note: This table reports the results of direct domestic value added and total domestic content share using different initial values. See section IV.D for details.

V. Conclusion

In most countries, statistical agencies compile an I/O table at the industry level for the entire economy. However, such I/O tables are not separately available for exporters and non-exporters. The I/O table assumes that only one single homogenous production technology exists for all of the firms (and all of the products) in the same industry category. As such, a single average production technology is regarded to produce the entire output of an industry. In reality, different firms, even those that produce the same products, often use different production technologies, and thus have different I/O coefficients and import intermediate use intensities. This paper proposes an estimation method for reducing the aggregation bias caused by firm heterogeneity in existing I/O tables by combined firm- and industry-level data, thereby making contributions to current vertical specialization and trade in value added literature. In addition, the proposed method can be potentially applied to other developed and developing countries.

To complement existing literature, we propose an extension to the approach of KWW (2012) by considering exports by both foreign invested enterprises and domestic-owned Chinese firms separately for processing exports and normal exports. Processing trade and FIEs play major roles in the fast growth of Chinese exports. However, they also use imported intermediate inputs much more intensively than normal exports by COEs. Thus, it is important to separate domestic production by trade mode and ownership to capture the processing exports and FIEs exports. By extending the theoretical framework of KWW (2012) and applying it to China's firm-level trade

and production data, we obtain more detailed estimates on domestic and foreign content shares in Chinese exports. Our empirical results show that the total domestic content share in gross Chinese exports is only 59 percent. Furthermore, processing exports and exports by FIEs have much lower domestic content shares. Within the processing export regime, the domestic content share at the sector level between FIEs and COEs differ across sectors, although the weighted sum is similar. This difference at the sector level justifies our categorization according to ownership and trade mode.

We further study the distribution of GNI by factor ownership. About 52.6 percent of the total Chinese export value is obtained by foreign factors owners. This finding implies that the Chinese local income from exports in value added term may not be as large as what official trade statistics have indicated.

References

[1] Ahmad, Nadim., and Sonia Araujo, (2011) "Using Trade Micro data to Improve Trade in Value Added Measures: Proof of Concept Using Turkish Data". OECD Working paper

[2] Brandt, L, J. Van Biesebroeck and Y. Zhang, (2011), "Creative Accounting or Creative Destruction? Firm-Level Productivity Growth in Chinese Manufacturing", *Journal of Development Economics*

[3] Branstetter, Lee, and Nicholas Lardy, (2008), "China's Embrace of Globalization", in Loren Brandt and Thomas Rawski, ed. "*China's Great Economic Transformation*", Cambridge University Press.

[4] Brooke, Kendrick, and Raman Meeraus, (2005), GAMS—User's Guide. GAMS Development Cooperation, Washington, DC.

[5] Chen, X., L. Cheng, K.C. Fung and L. J. Lau. 2004. "The Estimation of Domestic Value-Added and Employment Induced by Exports: An Application to Chinese Exports to the United States," *mimeo*, Stanford University.

[6] Chen, Xikang, Leonard Cheng, K.C. Fung, Lawrence Lau, Yun-Wing Sung, Cuihong Yang, Kunfu Zhu, Jiansuo Pei and Yuwan Duan (2012) "Domestic Value Added and Employment Generated by Chinese Exports: A Quantitative Estimation", *China Economic Review*, 23: 850-864.

[7] Dean, Judith, K.C. Fang and Zhi Wang, (2011) "Measuring Vertical Specialization: The Case of China", *Review of International Economics*, 19(4):609-625.

[8] Dedrick, Jason., Kenneth Kraemer, and Greg Linden (2009). "Who Profits from Innovation in Global Value Chains? A Study of the iPod and notebook PCs", *Industrial and Corporate Change* 19:1, 81-116.

[9] De la Cruz, Justin., Robert Koopman, Zhi Wang, and Shang-jin Wei, 2011 "Estimating Foreign valueadded in Mexico's manufacturing Exports." *Office of Economics Working Paper* No. 2011-04-A, U.S. International Trade Commission.

[10] Feenstra, Robert., and Gordon Hanson, 2005. "Ownership and Control in Outsourcing to China: Estimating the Property-Rights Theory of the Firm," *The Quarterly Journal of Economics*, vol. 120(2): 729-761.

[11] Feenstra, Robert., and Branford Jensen (2012), "Evaluating Estimates of Materials Offshoring from U.S. Manufacturing", *Economics Letters*, vol. 117(1):170-173.

[12] Hummels, David., J. Ishii and K. Yi. (2001), "The Nature and Growth of Vertical Specialization in World Trade," *Journal of International Economics* 54:75-96.

[13] Johnson, Robert and G. Noguera (2012), "Accounting for Intermediates: Production Sharing and Trade in Value Added", *Journal of International Economics*, 86 (2), 224-236.

[14] Kee, Hiau Looi, and Heiwai Tang (2012), "Domestic Value Added in Chinese Exports", Tufts University, mimeo.

[15] Koopman, Robert, Zhi Wang and Shang-jin Wei, 2012, "Estimating Domestic Content in Exports When Processing Trade is Pervasive", *Journal of Development Economics* 99: 178-189.

[16] Koopman, Robert, Zhi Wang and Shang-jin Wei, 2014, "Tracing Value-Added and Double Counting in Gross Exports", forthcoming in *American Economic Review*.

[17] Naughton, Barry (2006), "China's emergence and prospects as a trading nation". *Brookings Papers* on Economic Activity 2: 273-313.

[18] The Economist (2012), "iPadded", Jan 21st, 2012. http://www.economist.com/node/21543174.

[19] Upward, Richard, Zheng Wang, and Jinghai Zheng, (2012) "Weighing China's Export Basket: The Domestic Content and Technology Intensity of Chinese Exports", forthcoming, *Journal of Comparative Economics*

[20] Xing, Yuqing, and Neal Detert, (2011), "How the iPhone Widens the United States Trade Deficit with the People's Republic of China", ADB working paper No. 257